



OCCASION

This publication has been made available to the public on the occasion of the 50th anniversary of the United Nations Industrial Development Organisation.

TOGETHER

for a sustainable future

DISCLAIMER

This document has been produced without formal United Nations editing. The designations employed and the presentation of the material in this document do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations Industrial Development Organization (UNIDO) concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries, or its economic system or degree of development. Designations such as "developed", "industrialized" and "developing" are intended for statistical convenience and do not necessarily express a judgment about the stage reached by a particular country or area in the development process. Mention of firm names or commercial products does not constitute an endorsement by UNIDO.

FAIR USE POLICY

Any part of this publication may be quoted and referenced for educational and research purposes without additional permission from UNIDO. However, those who make use of quoting and referencing this publication are requested to follow the Fair Use Policy of giving due credit to UNIDO.

CONTACT

Please contact <u>publications@unido.org</u> for further information concerning UNIDO publications.

For more information about UNIDO, please visit us at www.unido.org



24 ×

5



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS STANDARD REFERENCE MATERIAL 1010a (ANSI and ISO TEST CHART No. 2)



HUMPHREYS & GLASGOW LIMITED

PREINVESTMENT STUDIES FOR THE PROMOTION OF THE FERTILIZER AND PETROCHEMICAL INDUSTRIES IN PARISTAN

a a

UNITED NATIONS INDUSTRIAL DEVELOPMENT ORGANISATION

 \vee () [U M F H

The Reputers David Managers and the the Reputers Reputer and Reputers Reputers and Reputers Reputers to the Reputers to the Reputers Reputers to the Reputers Reputer

2.2 Carliste Place London S. M. 1. England

PRE-INVESTMENT STUDIES FOR THE PROMOTION OF FERTILIER AND "
PETROCHEMICAL INDUSTRIES IN PAKISTAN



for

UNITED NATIONS INDUSTRIAL DEVELOPMENT ORCANISATION

VOLUME II

THE FURTHER DEVELOPMENT OF THE PETROCHEMICALS CONFLICTE PLANNED FOR EAST AND VEST PAKISTAN

July 1970

U.N.I.D.O. Pelderhaus, Rathausplats 2. A-1010 VIENNA. AUSTRIA

Humphreys & Glasgow Ltd.. 22. Carlisle Place. London, S.W.J. U.K.

Î

| | | Volume I |
|------------------------------|--|-------------------|
| HUPPPHREYS & C | Con | tents Pere No. |
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.166 July 197 |
| | | |
| | VOLUME II | |
| | THE FURTHER DEVELOPMENT OF THE PETROCHEMICALS COMPLEX | |
| | PLANNED FOR EAST AND WEST PAKISTAN | |
| Section Numbe | <u>CONTENTS</u> | Page Number |
| | Summary of Report | |
| Section 1 | INTRODUCTION & ACKNOWLEDGEMENTS | 1-1 |
| Section 2 | MARKET ESTIMATE | |
| 2.1 | Planned Production and Net Market Requirements for 1980 and 1985 | 2-1 |
| | TABLE II 2.1 | 2-2 |
| 2.2 | Fauji Foundation Phase I Production Quantities | 2-1 |
| 2.3 | Ashuganj Phase I Production Quantities | 2-3 |
| 2.4 | Other Existing or Santioned Capacity | 2-3 |
| 2.5 | Polyethylene | 2-3 |
| 2.6 | High Density Polyethylene | 2-3 |
| 2.7 | Polyvinyl Chloride | 2-4 |
| 2.8 | Polypropylene | 2-4 |
| 2.9 | Dioctyl Phthalate | 2-4 |
| 2.10 | Dodecyl Benzene | 2-5 |
| 2.11 | Polyvinyl Acetate | 2-5 |
| 2.12 | Vinyl Acetate | 2-6 |
| 2.13 | Polyvinyl Alcohol | 2-7 |
| 2.14 | Acetic Acid | 2-7 |
| 2.15 | Urea-Formaldehyde Resins | 2-7 |
| 2,16 | Demand for Intermediates Linked with Aromatics | 2-8 |
| • * . | Products | |
| 2.17 | Demand for End-Products or Intermediates Planned for Ashuganj Phase I | 2 -8 |
| Section 3 | FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPL | |
| 3.1 | Analysis of Development Schemes for Fauji Foundation Foundation, West Pakistan | 3-1 |
| | 3.1.1 Development Schemes Related to Fauji Foundation Future Plans | ons 3-1 |

P



| UNIDO Vienna Pre-Investment Studies for Fertiliser | 0.2 |
|---|-------|
| for Pakinton A Patrochemical Industries - Final Report | .1669 |
| ICF FARINCALL & FECTORIEMICAL INDUSTRIAN - FINAL Neport | 1970 |

ł

| Section Mumber | | CONTENTS | | Page Number | |
|----------------|---------------------|-------------------------|---|----------------|--|
| | | TABLE II | 3.1 | 3-1 | |
| | | FIGURE I | I 3.1 | 3-4 | |
| | 3.1.2 | The Prod Propylen | uction of Dioctyl Phthalate from e in West Pakistan | 3-2 | |
| | 3.1.3 | Consider | ation of Other End-Products | 3-3 | |
| 3.2 | Recommo Pau ji 1 | nded Deve Foundation | lopment and Financial Evaluation for , West Pakistan | 3-5 | |
| | 3.2.1 | Introduc | tion | 3-5 | |
| | 3.2.2 | Individu | al Process Units | 3-5 | |
| | | 3.2.2.1 | Dodecyl Bensene Plant | 3-5 | |
| | | | FIGURE II 3.2 | 3-6 | |
| | | | TABLE II 3.2 | 3-7 | |
| | | 3.2.2.2 | Polyethylene Plant | 3-8 | |
| | | | TABLE II 3.3 | 3- 8 | |
| | | 3.2.2.3 | Ethylene Plant | 3-8 | |
| | | | TABLE II 3.4 | 3-9 | |
| | 3.2.3 | Complete | Development Scheme | 3-9 | |
| | | 3.2.3.1 | Utilities and Offsites Installation | 3-9 | |
| | • | 3.2.3.2 | Utilities Consumption | 3-9 | |
| | | | TABLE II 3.5 | 3-10 | |
| | | 3.2.3.3 | Management, Personnel, Labour | 3-10 | |
| | | 3.2.3.4 | Plot Plan of the Recommended Development | N 3-1 0 | |
| | 3.2.4 | Capital | Costa | 3-11 | |
| | | 3.2.4.1 | Dodecyl Benzene Plant | 3-11 | |
| | | | TABLE II 3.6 | 3-11 | |
| | | 3.2.4.2 | Polyethylene Plant | 3-12 | |
| | | | TABLE II 3.7 | 3-12 | |
| | | 3.2.4.3 | Expansion to Phase II Sthylene Plant | 3-13 | |
| | | | TABLE II 3.8 | 3-13 | |
| | | 3.2.4.4 | Summary of Capital Coots | 3-14 | |
| | | | TABLE II 3.9 | 3-14 | |
| | 3.2.5 | Unit Ope | erating Costs | 3-15 | |
| | | TABLE II | 1 3.10 | 3-15 | |
| | 3.2.6 | Pereign | Exchange Analysis | 3-16 | |

| | | | | Volume I |
|--------------|------------------|--------------------------|---|---------------------|
| UNTIMEYS & | | LTD. | Conten | ts Page No. |
| UNIDO Vienna | L | Pre-Inv | estment Studies for Pertiliser | C.16 |
| for Pakistan | 1 | & Petroch | emical Industries - Final Report | July 19 |
| . | | | | |
| Section Rul | | _ | | |
| | | 3.2.6.1 | General | 3-16 |
| | | · | TABLE II 3.11 | 3-16 |
| | | 3.2.6.2 | Variable Costs of Production | 3-16 |
| | | | TABLE II 3.12 | 3-17 |
| | | 3.2. 6.3 | Fixed Costs of Production | 3-17 |
| | | , | TABLE II 3.13 | 3-19 |
| | | 3.2.6.4 | Foreign Exchange Saving and Rupee Cost Comparison TABLE II.3.14 | 3-19 3-20 |
| | 3.2.7 | 3.2.6.5 Cash Flo | Foreign Exchange Analysis for Each Pre w and D.C.F. Rate of Return | duct 3-23 3-22 |
| | | 3.2.7.1 | General | 3-22 |
| | | 3.2.7.2 | Ethylene and Propylene Costs | 3-22 |
| | | 3.2. 7. 3 | Variable Cost of Polyethylene and Dodecyl Benzene Production | 3-22 |
| | | 3.2.7.4 | Fixed Costs of Production | 3-23 |
| | | 3.2.7.5 | Other Fixed Costs | 3-23 |
| | | 3.2.7.6 | Cash Flows and D.C.F. Rate of Roturn | 3-24 |
| | | | TABLE II 3.15 | 3-25 |
| 3.3 | Cenelu | sions | | 3-26 |
| Section 4 | E.P.I. | D.C., EAST | PAKISTAN, PETROCHEMICAL COMPLEX | |
| 4.1 | Analys East P | is of Deve Wakistan | lopment Schemes for E.P.I.D.C., | 4-1 |
| | TABLE | II 4.1 | | 4-2 |
| | FICURE | : II 4.1 | | 4-4 |
| 4.2 | Necom Ashuga | ended Deve nj Site, B | lopment and Financial Evaluation for Last Pakistan | 4-5 |
| | 4.2.1 | Introduc | tion | 4-5 |
| | 4.2.2 | Individu | al Process Units | 4-5 |
| | | 4.2.2.1 | Acetaldehyde Plant | 4-5 |
| | | | FIGURE II 4.2 | 4-6 |
| | | | TABLE II 4.2 | 4-7 |
| | | 4.2.2.2 | 2 Ethyl Hexanol Plant | 4-7 |
| | | | FIGURE II 4.3 | 4-8 |
| | | | TABLE II 4.3 | 4-10 |

Y

| | / |) |
|---------------------|--|-----------------|
| | | Volume I |
| HUMPHREYS & GLASSON | V LTD. Cont | ents Page No. 4 |
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.166 |
| for Pakistan | & Petrochemical Industries - Final Report | July 197 |
| | | |
| Section Number | CONTENTS | Page Number |
| * | 4.2.2.3 Dicetyl Phthalate Plant | 4-10 |
| | FIGURE II 4.4 | 4-11 |
| | TABLE II 4.4 | 4-13 |
| | 4.2.2.4 Acetic Acid Plant | 4-13 |
| | FIGURE II 4.5 | 4-14 |
| | TABLE II 4.5 | 4-15 |
| | 4.2.2.5 Vinyl Acetate Plant | 4-26 |
| • | FIGURE II 4.6 | 4-17 |
| | TABLE II 4.6 | 4-18 |
| | 4.2.2.6 Acetylene Plant | 4-38 |
| | TABLE II 4.7 | 4-19 |
| 4.2.3 | Complete Development Scheme | 4-19 |
| | 4.2.3.1 Utilities and Offsites Installation | 4-19 |
| | 4.2.3.2 Utilities Consumption | 4-20 |
| | TABLE II 4.8 | 4.20 |
| | 4.2.3.3 Management, Personnel, Labour | 4-20 |
| | 4.2.3.4 Plot Plan of the Recommended | 4-20 |
| | Development Frankr TI & 7 | A_91 |
| | 4.2.3.5 Use of Acetylene Off-gas | 4-82 |
| 4.2.4 | Capital Costs | 4-24 |
| | 4.2.4.1 Acetaldehyde Plant | 4-24 |
| | TABLE II 4.9 | 4-24 |
| | 4.2.4.2 2 Ethyl Hexanol Plant | 4-25 |
| | TABLE II 4.10 | 4-25 |
| | 4.2.4.3 Dioctyl Phthalate Plant | 4-26 |
| | TABLE II 4.11 | 4-26 |
| | 4.2.4.4 Acetic Acid Plant | 4-27 |
| | TABLE II 4.12 | 4-27 |
| | 4.2.4.5 Vinyl Acetate Plant | 4-28 |
| | TABLE II 4.13 | 4-28 |
| | 4.2.4.6 Expansion to Phase II Acetylene Plan | t 4-29 |
| | TABLE II 4.14 | 4-29 |
| | 4.2.4.7 Capital Costs Summary for Expansion | 4-30 |
| | TABLE II 4.15 | 4-30 |
| | 4.2.4.8 Pre-Investment for Phase II | 4-31 |
| | 4.2.4.9 Capital Cost Summary for Project | 4-51 |

P

Volume II

HUMPHREYS & GLASSOW LTD.

Contents Page No. 5

| Tor registeri & retrochemical industries - Final Meport | July 1970 |
|---|------------------|

| Section Number | 2 | | CONTENTS | Page Number |
|----------------|----------|----------|---|--------------|
| • • • | • | | TABLE II.4.16 | 472 |
| | 4.2.5 | Unit Ope | rating Costs | 4-33 |
| | | • | TABLE II.4.17 | 4-33 |
| | 4.2.6 | Poreign | Exchange Analysis | 4-34 |
| | | 4.2.6.1 | General | 4 -34 |
| | | | TABLE II 4.18 | 4-34 |
| | | 4.2.5.2 | Variable Costs of Production | 4-34 |
| | | | TABLE II 4.19 | 4 -34 |
| | | 4.2.6.3 | Fixed Costs of Production | 4-35 |
| | | | TABLE II 4.20 | 4-36 |
| | | 4.2.6.4 | Foreign Exchange Saving and Rupse Cost Comparison | 4-36 |
| | • | 4.2.6.5 | TABLE II.4.21 Romain Exchange Analysis for Rock Product | 4-37 |
| | 4.2.7 | Cash Flo | w and D.C.F. Rate of Return | 4-30 4-40 |
| | | 4.2.7.1 | General | 4-40 |
| | | 4.2.7.2 | Average Cost of Acetylene | 4-40 |
| | | 4.2.7.3 | Variable Cost of Dioctyl Phthalate, Acetic Acid, and Vinyl Acetate | 4.40 |
| | | 4.2.7.4 | Fixed Costs of Production | 4_41 |
| | | 4.2.7.5 | Other Fixed Costs | 4-41 |
| | | 4.2.7.6 | Cash Flows and D.C.F. Rate of Return | 4.42 |
| | | | TABLE II 4.22 | 4.43 |
| 4.3 (| Conclusi | ons | | 4.44 |
| | Appendix | I | Bibliography | |
| In pocke | t | | Drg. No. C.1669 L 51 - 4 | |

| HUMPHREYS & GLAS | GOW LTD. | Volume II |
|------------------|---|-----------|
| UNIDO Vienna | Pre-Trivestment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |
| | | |

VOLUME II

THE FURTHER DEVELOPMENT OF THE PETROCHEMICALS COMPLEXES PLANNED FOR EAST AND WEST PAKISTAN

Summary of Report

The Report studies the possible new products that could be manufactured at Ashuganj, East Pakistan and Korangi, West Pakistan. It establishes the market requirements additional to production from existing and sanctioned plants, including the Phase I developments at both of these sites.

In the preparation of plans for the second stage in the development of the two projects. the current production schemes envisaged by E.P.I.D.C. and Fauji for Phase I of their projects were used as the basis. The products and capacities recommended are also related to the future market requirements, suitability of the sites for manufacture of the products concerned and the scale of production needed for economic viability. The specific developments recommended are:-

East Pakistan: Abhuganj (E.P.I.D.C.)

Increase acetylene production from 36,000 MTPA to 55,000 MTPA Produce: - 16,000 MTPA dioctyl phthalate 8,420 MTPA vinyl acetate 1,650 MTPA acetic acid

West Pakistans Korangi (Fauji Foundation)

Increase ethylene production from 30,000 MTPA to 60,000 MTPA (already planned) Produce:- 23,600 MTPA polyethylene, low density 10,000 MTPA dodecyl bensene 5,300 MTPA ethylene for sale to Valika

After establishing the capital and operating costs of the individual plants and recommended developments as a whole, the Rupse cost of saving foreign exchange and the Cash Flow (including the D.C.F. rate of return on the Rupse investment) were studied. These were found to be:

East Pakistan, Ashuganj

| Cost of Saving | Foreign Exchange | Rs. 5.8 U.S. S |
|----------------|----------------------------|----------------|
| D.C.F. Rate of | Return on Rupee Investment | 16.45 |

| HUMPHREYS & GLAD | 60W L7D. | Volume II |
|------------------|---|-----------|
| UNTDO Vienna | Pre-Investment Studies for Pertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

West Pakistan, Korangi

Cost of Saving Foreign ExchangeRs. 4.38/U.S.#D.C.F. Rate of Return on Ruppe Investment43.86

It was concluded that the cost of saving foreign exchange was favourable for both East and West Pakistan. In East Pakistan, the D.C.F. rate of return was considered acceptable, but its improvement by delaying start-up after 1976 was suggested. In West Pakistan, the economic aspects of the project are so favourable that an earlier start-up date than 1976 could well be considered by Pauji Foundation.

The Report also provides process descriptions, flow charts, and operating data for the individual processes.

HUMPHREYS & GLASGOW LTD.

UNIDO Vienna for Pakistan Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report C.1669

Volume II

1 - 1

July 1970

VOLUME II

THE FURTHER DEVELOPMENT OF THE

PETROCHEMICALS COMPLEXES CURRENTLY PLANNED FOR

LAST AND WEST PAKISTAN

SECTION I

INTRODUCTION AND ACKNOWLEDGEPHENTS

At present there are two major petrochemicals complexes projected for Pakistan.

The Fauji Foundation are in the final stages of contract negotiations for their naphtha-based, PVC/Polyethylene complex at Korangi, near Karachi. EPIDC is in the stage of detailed planning for a project at Ashuganj based on natural gas and acetylene.

In each case the projects are being planned in two phases. While the production units envisaged for phase I of each project are now well defined, the phase II plant units have yet to be specified. This report is concerned with the specification of the phase II projects. The market data upto 1985, so far as they exist, have been analysed and products for phase II manufacture have been selected. These have been incorporated in an overall production pattern. The benefits of the scheme have been analysed.

The production schedules for phase I of the two projects are as follows:

(a) <u>Fauli Foundation</u>, Korangi

| | Ethy) | lene(to Valika) | 5,800 |
|-----|--------------------------|----------------------|---------|
| | P.V. | . | 15,000 |
| | Poly | rthylene | 15.000 |
| | Poly | propylene | 10,000 |
| | Estimated capital cost 1 | ts 344 million (Cher | ncon)· |
| (b) | EPIDC, Ashuganj | | MTPA |
| | Acety | lene | 3,600 |
| | P.V.C | € * ● | 50,000 |
| | Acryl | ic Fibres | 16,000 |
| | Methy | 1 Methacrylate | 2,600 |
| | Polym | ethyl Methacrylate | 4,000 |
| | Caust | ic Soda | 42,000 |
| | Annon | ium Sulphate | 28,000 |
| | Bleac | hing Powder | 1,000 |
| | Metha | nol | 37,000 |
| | Urea | | 320.000 |

| HUMPHREYS & GLAS | GOW LTD. | Volume T. 1-2 |
|------------------------------|--|------------------|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 |

In the Interim Report prepared by H & G in lovember 1969 and now summarised as Volume (of this report, two products additional to those above were recommended for production in Pakistan. These are dodecyl hanzene. required as a detergent precursor, and diorty! phthalate. a plasticiser used with PVC. These have, therefore, been included in our recommendations for phase II of the Fauji and EPIDC projects.

No allowance has been made for possible exports of those products considered for the extensions to the two petrochemical complexes. The reasons for excluding exports are a) the basis of H & G's study is primarily the Pakistan market, and b) the particular products considered could probably not be produced at a cost which would justify extra investment based on sales over a long period in the world market. Exports should, however, be attempted during the years 'up to about 1980) when domestic requirements are lagging behind production capacity. It should be possible to achieve a certain amount of such sales at prices above total variable cost, which would improve the economics and profitability of the proposed extensions beyond the figures quoted in this report. It is recommended that a survey of export possibilities be undertaken.

The basis for assessment of additional production has been the total Pakistan market, the initial part of our study, culminating in the Interim Report. having shown that it is generally more attractive to make products required on such a scale in a single plant serving both Wings. We have also taken into account any potential requirements for use in the aromatics based industries proposed in Volume III.

In this Volume, cost analysis of the projects is made in two distinct ways. In the analysis of foreign exchange savings costs and benefits are expressed in terms of estimated effects on the economy of Pakistan as a whole. For this reason, duty payments and cost of bonus vouchers are excluded from the cost of imported items and bonus value from the credit value of exports. To present this information in the most useful form, we have kept foreign exchange and local currency costs separate. looking at the petrochemical industry as a whole. The ratio of rupees expended to foreign currency 'expressed 'n dol'ars' is then calculated.

Commercial profitability is analysed separately on a cash flow basis. In calculating the cash flows.Foreign exchange is converted to rubees at the official rate, and duty and bonus payments are included. Foreign exchange loan repayments and interest, and interest on working capital, are treated as cash outflows from the project. The DCF rate of return is calculated from the resulting cashflows, which is equivalent to treating all rupee finance (other than for working capital) as equity.

....

Humphreys & Glasgow Ltd. wish to express their appreciation of the help and information given to them by:

The Government Project Representative in Rawalpindi. the U.N. I.D.O. Project Manager, and Co-Project Manager, and U.N.I.D.O. Experts.

| HUNPHREYS & GLASG | OW LTD. | Volume II 2-1 |
|------------------------------|--|------------------|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Pertiliser & Petrochemical Industries - Final Report | C.1669 |

SECTION 2

MARKET ESTIMATES

2.1 Planned Production and Net Market Requirements for 1980 and 1985

Table 11.2.1 (p.2-2) summarises the production of petrochemicals planned for Phase I of the Petrochemicals Complexes in both Wings. This, together with other sanctioned or existing production and the market demand quantities for 1980 and 1985, gives, by difference, the net additional market requirements for these two years, which are then considered in later sections of the report.

The product list is, as described in Section 1, extended beyond that applicable to Phase I production.

Market demand data for 1980 is, with a few exceptions discussed later in this section, taken from our Interim Report. Market demand data for 1985 is taken very largely from the Battelle Institute Report. Battelle's information has been analysed and correlated with data provided by UNIDO documents and by Chemical Consultants Ltd.

2.2 Fauji Foundation Phase I Production Quantities

*1

The quantities listed in column 1 of Table 11.2.1 for Phase I of the Fauji Project are based on an initial Phase I cracker capacity of 30,000 MTPA, and a material balance on this basis establishes that there will be, approximately, surpluses of propylene and ethylene to the following extent:

| Propylen | • | | | | 1900 | MTPA |
|----------|-----|------|----|--------|------|------|
| Ethylene | for | sale | to | Valika | 5500 | MTPA |

Surplus propylene, which is not for the present purposes defined as an end product. may either be burnt as fuel or at least partially utilised if the capacity of the polypropylene plant exceeds its nominal design rate.

In the previous work for the Interim Report, a sale of 10,500 MTPA of ethylene to Valika and a sale of 5,500 MTPA of vinyl chloride to Arokey were envisaged. The sale of ethylene to Valika is still probable, whereas we are advised that the sale of vinyl chloride to Arokey now seems unlikely. Accordingly, it is assumed that there will be no sales of vinyl chloride monomer, and that a total of 10,500 MTPA of ethylene will be sold to Valika when Phase II is in full production. This implies that Valika will continue to operate their ethylene plant to make up the balance of their requirements, amounting to about 5,000 MTPA of ethylene from molasses, at their full sanctioned capacity of 15,000 MTPA of polyethylene.

| | Andres a California a Californi | | | Pre- Pet: | Cochem Co | tment ical | Studi Indust | es for ries - | Perti Final R N | lise Rep Rep | r ort 9 9 8 | | Volume C.1 July 1 |
|--|--|-------------|--|---|--|---------------|-----------------|------------------|--------------------------|--------------------|-------------------------|--------------|---|
| RODUCTI | 1990 Net May Medicional to WIDC and oil Smertioned Ca | 1 S | 800 | 8 | 88 | 8 | 98 X | 8 | 8 | ŝ | 95 | R | lathacrylate a |
| CALDP | | ž A T | 0009 | | 80049 | 900 9X 90 | | 11,000 | 8 | 9 2 2 | 92.01 8 | 154.00 | , 200 methyl innet to satis |
| Q U I R E M E M I les are expressed | | ž | 10 50 50 50 50 50 50 50 50 50 50 50 50 50 | 9 8 7 | | 9655 9025 | ¥7 005 0002 | | | | | 6100 6100 | L.T.P.A. aortyleen Mities are suffic |
| AKETRE (All quantit | datie/ | 1 | | * .*********************************** | | | | | 8 | | + | | These gues |
| L S - H A | | Ĭ | 15000 | | 9995 | | | | 8 | | | 8 | des else i rylic fibre. |
| | | | | | | | | | | | | | 1 protection |
| | 1 | įį | 15000 | | and the | 9909 | | | | | | | |
| | 1 | | المالية الم | K. D. bijyethy lame | Pulyring I Chloride | Pigerog las | Diantyl |]] | A Lyning L | tinyl hotato | testie Acid | | Mandas Britic A |

,

1

٢ 1.

| H | umphreys a glasg | OW LTD. | Volume I . 2- |
|--------|---|---|--|
| U f | UNIDO Vienna For Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.166 July 197 |
| | | | |
| 2 | .3 Ashuganj Ph | ase I Production Quantities | |
| | The E.P.I.D.C.' January 197 chemical Co complex, pr diagram 2 (that diocty have accord and reliabl | quantities stated in column 2 of Table II.2.1 were s proposal for a petrochemical complex in East Paki 0 which in turn was based very largely on the Austr nsultants Ltd. investment study of an East Pakistan epared for the E.P.I.D.C. The quantities were tak Simplified Flow Chart) of the E.P.I.D.C. proposal. 1 phthalate is not now envisaged for Phase I at Ash ingly used the E.P.I.D.C. data as the most complete e representation of E.P.I.D.C.'s plans. | based on stan, dated ian Petro- petrochemical en from It is clear uganj, and we , up-to-date, |
| | A s acetylene c acetylene f producing 4 surplus of for the man | imple material balance of the E.P.I.D.C. scheme con apacity of 36,000 MTPA (the only discrepancy being or direct sales). The only other discordant fact 2,000 MTPA caustic soda - as stated by E.P.I.D.C., approximately 5,000 MTPA chlorine in addition to it ufacture of bleaching powder. | firmed the bas 400 MTPA more was that, when there was a s utilisation |
| | The ducts, such Phase I pro petrochemics will proceed H. & G. in T | products covered by Table II.2.1 do not include so as urea, bleaching powder, and ammonium sulphate, duction at Ashuganj, as they arise as by-products in als plant. It now seems likely that the urea plan i independently of the petrochemicals project as rea the Interim Report. | me of the pro- planned for n the E.P.I.D.(t at Ashuganj commended by |
| 2 | .4 Other Exist | ing or Sanctioned Capacity | |
| | The from our In these in de | quantities stated in columns 3 and 4 of Table II.2 terim Report. The Battelle Institute Report discustail. | l are taken sees many of |
| | Fur in the subs | ther discussion of the existing or sanctioned capac: equent sections for the individual petrochemical en | ity is included i products. |
| 5 | .5 Polyethylen | 2 | |
| | The further 10,0 | existing production of polyethylene at Valika is 5, 000 MTPA is sanctioned, making a total of 15,000 MT | ,000 MTPA and a PA in all. |
| | The The 1985 man same distril | 1980 market demand is based on the figures in our in rket demand is taken from the Battelle Institute Rep oution of demand between the two Wings as for 1980. | Interim Report, port, with the |
| | The | net additional requirements are obtained by different | ence. |
| 2 | .6 <u>High Densit</u> | Polyethylene | |
| | The: Pakistan | re is no existing or sanctioned production capacity The 1980 and 1985 market demands have been taken fo | at present in |

| Ľ | HUMPHREYS & GLASS | SOW LTD. | Volume II 2-4 |
|---|-------------------|---|------------------|
| Γ | UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

2.7 Polyvinyl Chloride

لده

٦

1

1

j

ą.

Π

The present production of polyvinyl chloride at Arokey is 5,000 MTPA and future extensions to their existing capacity are uncertain. The existing/sanctioned capacity has only, therefore, been stated at 5,000 MTPA, for West Pakistan. At present no polyvinyl chloride is produced in East Pakistan.

The 1980 market demand is based on the figures in our Interim Report. Those for 1985 are taken from Battelle Institute with the same distribution of demand between the two Wings as for 1980. Although the demand in East Pakistan is expected to increase at a greater rate than in West Pakistan over the period 1968 to 1975, as the Battelle Report points out, this picture may well change if PVC pipes are used in the public sector of West Pakistan to the same extent as in East Pakistan. In addition 1985 is so far ahead that even the total demand figures must be viewed as being relatively less reliable than those for earlier years.

The net additional requirements for the two years are obtained by difference and it should be noted that, according to this reckoning, polyvinyl chloride will be overproduced in East Pakistan even in 1980 as far as the home market is concerned. Even allowing for the requirement of 8,000 MTPA in 1980 in West Pakistan, and the necessary interwing transfer, there will still be a surplus of 10,000 MTPA for which an export market should be found. The surplus in earlier years will be greater assuming that the plant is to be utilised to the maximum extent.

2.8 Polypropylene

There is no existing or other sanctioned production in Pakistan, and the 1980 market demands are taken from our Interim Report. The 1985 market demands are taken from the Battelle Institute Report, and, as with polyethylene and PVC, the distribution of demand between the two Wings has been taken as for 1980.

2.9 Dioctyl Phthalate

Dicety phthalate is not at present produced in Pakistan, and there is no production for Pakistan sanctioned.

At Ashuganj, the production of dioctyl phthalate for Phase I has been considered, but according to the latest information available to H. & G. it will be excluded. The production of this petrochemical end product was recommended in our Interim Report, and further consideration still shows it to be economically profitable both from the national and commercial points of view.

The 1980 market demands are taken from our Interim Report. The 1985 market demands are based on the increased demands for PVC shown in Table II.2.1 maintaining the same proportion of plasticiser to polymer. It must be noted, however, that these figures may well be high, as a result of the increasing use of the more rigid form, such as PVC pipes. The distribution of demand between the two Wings has been taken as for 1980.

| HUMPHREYS & GLAS | SOW LTD. | Volume II 2-5 |
|------------------|---|------------------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

2.10 Dodecyl Benzene

There is no existing production of dodecyl benzene in Pakistan, and, apart from Phase I of the Fauji complex to be erected at Korangi, no other sanctioned production either.

At Korangi, the production of 5,000 MTPA of dodecyl benzene has been considered for Phase I, with an extension to 10,000 MTPA for Phase II. According to the latest information available to H. & G., Fauji now intend to delay the installation of the dodecyl benzene plant until Phase II, and instal a plant with a capacity of 10,000 MTPA.

The production of dodecyl benzene was recommended in our Interim Report and further consideration has not altered this recommendation.

The 1980 market demands are taken from our Interim Report. The Battelle Institute Report gives market demands for dodecyl bensene sulphonate, and for 1980 these are equivalent to our estimate. This report has been used as the basis for the 1985 market demands, with the same split between the two Wings as in 1980.

2.11 Polyvinyl Acetate

1

The Interim Report gave the estimated market demands for polyvinyl acetate in 1980 as 3,400 MTPA and the existing plus sanctioned capacity as 3,100 MTPA. Both quantities refer to emulsion, containing between 40% and 60% of polymer. This was based mainly on the Battelle Institute Report. The very small incremental requirement was not considered to be economically viable, and production was not recommended.

Two of the U.N.I.D.O. references, vis:-

Reference 1 "General Survey of the Existing Situation in the Petrochemical and Fertiliser Industries (C. Jenic)."

Reference 44 "Proposal of Feasibility Pattern of Petrochemical Complex Industry in Pakistan. Volume III (T. Janakievski)."

both indicated that the 1980 market demands were approximately 8,000 MTPA, but gave no evidence in support of this conclusion.

The Battelle Institute Report indicates that the main uses of polyvinyl acetate will be for the manufacture of paints, textile finishing, and adhesives. For 1975 these uses were expected to account for 76%, 22% and 2% of the total usage respectively.

An independant survey by Chemical Consultants Ltd. ("The Manufacture of Petrochemicals in East Pakistan" Volume II, July 1969.) indicates that the requirements for 1975 are 960 MTPA of polymer and for 1980 are 3,000 MTPA of polymer. Briefly, these figures are based on 20% and 30% respectively of the total paints production being of the emulsion type.

| | | Volume II |
|------------------|---|-----------|
| HUMPHREYS & GLAS | SOW LTD. | 2-6 |
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

The use for textile sizes, in the form of polyvinyl alcohol produced from polyvinyl acetate, is dependent on the growth of the textile industry, and specifically of synthetic fibres and fine cottons. The requirement for textile sizes in both Wings for 1980 is conservatively estimated to be about 30,000 MTPA, expressed as starch. Of this quantity, a conservative 7% is estimated to be required for synthetics and fine cottons, viz. 2100 MTPA, expressed as polyvinyl alcohol. The equivalent quantities of vinyl acetate required would be 4,000 MTPA.

A further use not mentioned in the Battelle Institute Report is for the manufacture of polyvinyl chloride/polyvinyl acetate co-polymer floor tiles. This is expected to give rise to a demand for about 300 MTPA of vinyl acetate in 1980.

The results of our further investigations are summarised below in Table II.2.2

TABLE II.2.2

Expected 1980 Market Demand for Polyvinyl Acetate

| Rnd Use | MTPA |
|--|------------------------------------|
| Paints Textile Finishes PVC/PV Acetate Floor Tiles | 3000 4700 30 0 |
| TOTAL | 8000 |

This revised 1980 market demand is in agreement with the two U.N.I.D.O. references mentioned earlier in this section, and is now considered to be more realistic than that given in our Interim Report. The split between the two Wings given in Table II.2.1 is based mainly on the Battelle Institute Report. Market demands for 1985 have not been investigated in the same detail, although an average growth rate of 15% per annum as used by Battelle, gives an approximate guide.

The existing and sanctioned capacity for the production of polyvinyl acetate emulsions is taken from the Battelle Institute Report. The existing capacity is 600 MTPA at Karachi (Futehally Chemicals Ltd.). Sanctions have been granted for 1,000 MTPA capacity at Karachi (Hoschst Chemicals) and for 500 MTPA capacity at Dacca (Pakbay Co.). In addition, Shuja Industries are reported to be planning a capacity of 1,000 MTPA at Karachi. It is assumed that the polymer content of the emulsion is 50% in all cases.

2.12 Vinyl Acetate

The small quantities of vinyl acetate polymerised at present in Karachi (15 tons

F



Volume II 2-7

| UNTDO Vienna | Pre-Investment Studies for Fertiliser | C. 1669 |
|--------------|---|----------------|
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

in 1967, and an estimated 35 tons in 1968) are imported.

The vinyl acetate requirement corresponding to the total market demands of polyvinyl acetate (see Section 2.11) are given in Table II.2.1 and split between the two Wings in the same ratic as the demand for polyvinyl acetate.

Vinyl acetate is included in Table II.2.1 because the manufacture of this product is recommended whereas the polymerisation to polyvinyl acetate is not recommended at Ashuganj. It is considered that the additional polymerisation facilities will be provided by extension of the existing facilities at the "point of usage", where the individual plant capacities will still be relatively small.

2.13 Polyvinyl Alcohol

1

1

Ť

The requirements for this end product are included in requirements for polyvinyl acetate (see Section 2.11) and are not listed separately in Table II.2.1.

The conversion of polyvinyl acetate to polyvinyl alcohol (approximately 1800 MTPA of the alcohol) is on a very modest scale and should be closely integrated with the polymerisation of vinyl acetate.

2.14 Acetic Acid

In addition to the requirements of vinyl acetate production for acetic acid, there is an additional demand for acetic acid itself. The 1980 market demand has been taken from our Interim Report and is exclusive of any acid produced by the fermentation route.

The relative demand for the two Wings is assumed to be equal. The 1985 market demand is based on an average growth rate of 65 in line with that used by the Battelle Institute for ethanol.

2.15 Urea-Formaldehyde Resins

Existing and sanctioned capacity for the production of ureaformaldehyde resins is taken from our Interim Report. The figures are in terms of dry resin, and existing production is centred in Karachi and Chittagong. The urea and formaldehyde are purchased from internal sources of supply. The sanctioned capacity at Jhelum in West Pakistan is included.

The 1980 market demands are also taken from our Interim Report, and the split between the two Wings is based on information contained in the Battelle Institute Report. The 1985 market demands are taken from this report also.



Volume II 2-8

HUMPHREYS & GLASGOW LTD.

1

UNIDO ViernaPre-Investment Studies for Fertiliser0.3669for Pakistan& Petrochemical Industries - Final ReportDuly 1970

2.16 Demand for Intermediates linked with Aromatics Products

These include: -

- (a) <u>Butadiene</u> required for synthetic rubbers. Butadiene may be extracted from the Fauji ethylene plant C_{ij} stream, which contains 39% ^W/w for Phase I operation and 55% ^W/w for Phase II operation. The manufacture of synthetic rubbers, and the extraction of butadiene is considered in Volume III of this Report.
- (b) <u>Ethylene Glycol</u> required for polyester fibres. This intermediate is not required in quantities sufficient to justify its manufacture in Phase II of the Fauji project, and its supply is discussed in further detail in Volume III of this Report.
- (c) <u>Styrene</u> required for S.B.R. rubbers and polystyrene. The manufacture of these two end products is considered in Volume III of this Report. The quantity of styrene required is inadequate to justify manufacture in Pakistan within the timescale under consideration.

2.17 Demand for End-Products or Intermediates Planned for Ashuganj Phase I

These consist essentially of :-

Methyl methacrylate Polymethyl methacrylate

In the case of polymethyl methacrylate, our estimate of the market demand for 1980 is based on the finding of the Battelle Institute Report, viz. 1800 MTPA. This scale of production would not have led us to recommend its manufacture for Phase I of petrochemical development in Pakistan and the planned E.P.I.D.C. capacity of 4,000 MTPA for Phase I is considered quite adequate to meet the demands up to and beyond 1980.

The same remarks apply to the production of the monomer, where it is noted that some of the Phase I production at Ashuganj is to be used in the finishing of the planned acrylic fibres production.

The demand for intermediates (acetone, acetone cyanhydrin, and acrylonitrile) required for these two products, and also acrylic fibres, has not, therefore, been considered any further than in our Interim Report.

The planned production rates for methyl methacrylate and polymethyl methacrylate, have been noted in Table II.2.1. In the case of acetylene, some is to be sold in pressurised bottles on the market. Other nonpetrochemical products have been excluded from Table II.2.1

| UNIDO Vienna Pre-Investment Studies for Fertiliser C.16 for Pakistan & Petrochemical Industries - Final Report July 19 SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX 3.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes for Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS | NIDO Vienna Pre-Investment Studies for Fertiliser C.16 & Petrochemical Industries - Final Report July 19 SECTION 3 SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX S.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan J.1.1 Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. Image: Addition of the faugi foundation of the faugi foundation for the purposes of this study. Image: FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS | HUMPHREYS & G | LAIGOW LTD. | | |
|---|---|---|--|--|---|
| SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX 3.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> <u>FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS</u> | SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX 5.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 5.1.1 Development Schemes related to Fauji Foundation's Future Plane The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TAELE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS (PHASE I) | UNIDO Vienna for Pakistan | Pre-Investment & Petrochemical] | Studies for Fertiliser Industries - Final Report | C.16 t July 1 9 |
| SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX 3.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> <u>FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS</u> | SECTION 3 FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS <u>MTPA</u> <u>Remarks</u> | | | | |
| FAUJI FOUNDATION, WEST PAKISTAN, PETROCHEMICALS COMPLEX 3.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan J.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> <u>FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS</u> | 5.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 5.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS <u>MTPA</u> <u>Remarks</u> | | 5 | SECTION 3 | |
| 3.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> <u>FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS</u> | 5.1 Analysis of Development Schemes for Fauji Foundation, West Pakistan 3.1.1. Development Schemes related to Fauji Foundation's Future Plans The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS <u>MTPA</u> (DHASE I. 1) | 1 | AUJI FOUNDATION, WEST | PAKISTAN, PETROCHEMICAL | S COMPLEX |
| The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> <u>FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS</u> | The further development of Phase I of the Fauji Foundation petrochemical complex has been considered in the light of Fauji's tentative plans for Phase II. Details of these tentative plans were obtained from Chemical Consultants Ltd., who were authorised by Fauji to disclose the information for the purposes of this study. <u>TABLE II.3.1</u> FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS (PHASE T & II) | | | | |
| TABLE II.3.1 FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS | TABLE II.3.1 FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS (PHASE T & II) | 3.1 <u>Analysis</u> 3.1.1. <u>De</u> | of Development Schemes evelopment Schemes rela | s for Fauji Foundation, W ated to Fauji Foundation | Vest Pakistan 's Future Plans |
| FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS | FAUJI FOUNDATION'S TENTATIVE PHASE II PLANS | 3.1 <u>Analysis</u> 3.1.1. <u>De</u> Tr pe Fr to wt th | of Development Schemes evelopment Schemes relative trochemical complex has auji's tentative plans intative plans were obtain to were authorised by H the purposes of this stu | s for Fauji Foundation, w ated to Fauji Foundation of Phase I of the Fauji as been considered in the for Phase II. Details tained from Chemical Cons Fauji to disclose the ins ady. | Vest Pakistan 's Future Plans Foundation a light of of these sultants Ltd., formation for |
| | (DHASE T. TT) | 3.1 <u>Analysis</u> 3.1.1. <u>De</u> Tr Pe Fa ta vit | of Development Schemes evelopment Schemes relative trochemical complex has auji's tentative plans intative plans were obtain the purposes of this stu- <u>Trin</u> | s for Fauji Foundation, Mated to Fauji Foundation of Phase I of the Fauji as been considered in the for Phase II. Details tained from Chemical Cons Fauji to disclose the ini- ady. ABLE II.3.1 | Vest Pakistan 's Future Plans Foundation a light of of these sultants Ltd., formation for |
| Total Ethylene 60.000 Twice Phase T | | 5.1 <u>Analysis</u> 3.1.1. <u>De</u> Tř Pi te vř tř | of Development Schemes evelopment Schemes rela the further development strochemical complex ha auji's tentative plans intative plans were obtain the purposes of this stu- <u>T/</u> <u>FAUJI FOUNDATIONE</u> Stal Ethylene | s for Fauji Foundation, we have ated to Fauji Foundation of Phase I of the Fauji as been considered in the for Phase II. Details tained from Chemical Constant of the fauji to disclose the instand. ABLE II.3.1 ON'S TENTATIVE PHASE II I MARKE I & II.3.1 MTPA (PHASE I & II) 60.000 | Vest Pakistan 's Future Plans Foundation a light of of these sultants Ltd., formation for PLANS Remarks Twice Phase I |
| Total Ethylene60,000Twice Phase IPolyvinyl Chloride15,000No changePolyvinyl Chloride10,000No change | Polyvinyl Chloride 15,000 No change | 5.1 <u>Analysis</u> 3.1.1. <u>De</u> Tr Pi te with th | of Development Schemes velopment Schemes rela the further development strochemical complex ha auji's tentative plans intative plans were obtain the purposes of this stu- <u>The FAUJI FOUNDATION</u> Stal Ethylene Slyvinyl Chloride | s for Fauji Foundation, we have at the fauji Foundation of Phase I of the Fauji as been considered in the for Phase II. Details tained from Chemical Constant of the fauji to disclose the instant of the fault of the fa | Vest Pakistan 's Future Plans Foundation a light of of these sultants Ltd., formation for PLANS Remarks Twice Phase I No change |

Dodecyl Benzene

.

The developments considered have, therefore, been restricted to a total ethylene capacity of 60,000 MTPA. No further expansion of the PVC production capacity has been considered as the Pakistan Government has given sanction to the production of PVC via the acetylene based route in East Pakistan using indigenous supplies of natural gas. This is reflected in Fauji's tentative plans above. Further, as stated in Section 2.7, according to our market demand estimates, PVC will be overproduced for Pakistan as a whole until beyond 1980.

10,000

New product

The possible developments have therefore resolved themselves into the alternative ways of utilising the surplus ethylene and propylene produced from a 60,000 MTPA ethylene cracker, excluding any expansion of PVC.

High severity cracking conditions, which maximize the ethylene yield and minimize the propylene yield, are necessary for 60,000 MTPA of ethylene and this provides sufficient surplus ethylene to increase the production of polyethylene from 15,000 MTPA to 38,600 MTPA, after providing for the transfer of 10,500 MTPA of ethylene to Valika. The increase in polyethylene production is 23,600 MTPA and is still less than the 1980 net additional requirement of 32,000 MT for both wings shown in Table II.2.1 (which takes

| J | HUMPHREYS & GLA | GOW LTD. | Volume II 3-2 |
|---|-----------------|---|------------------|
| | UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

into account the total output from Valika as well as that from Fauji Phase I). Only 10,000 MTFR of this increase in polyethylene production will be required for the West Pakistan market in 1980.

Including inter-wing transport of the remainder to East Pakistan, however, the Interim Report indicated that the foreign exchange savings resulting from the manufacture of polyethylene were much greater than from the manufacture of polypropylene or dodecyl benzene, and the choice of cracking conditions which maximises ethylene production is therefore justified on this score alone.

Excluding the manufacture of dloctyl pathalate via Oxo Synthesis, the surplus propylene may be used either to manufacture additional polypropylene or start the manufacture of dodecyl benzene. In view of the greater uncertainty in the estimated market demands for polypropylene, and the higher foreign exchange savings indicated in our Interim Report for dodecyl benzene (over twice as much per M.T. of product), it was considered that the surplus propylene should be utilised entirely for the manufacture of dodecyl benzene.

The Fauji Foundation are understood to have fairly firm plans for the production of 10,000 M T P.A. of dodecyl benzene and the final recommended scheme has, in this case, been adjusted to coincide with their plans. This does, however, leave a surplus of 1,300 MTPA of propylene. Slight increases in the severity of cracking may be possible in practice to reduce this surplus, which will otherwise be used as fuel. Alternatively, Fauji may make the necessary 10% increase in the capacity of the DDB unit.

The Block Diagram II.3.1 summarises these conclusions and depicts our recommended development for this site. In addition to the main petrochemical end products, by-products have also been shown at the foot of the diagram. These have been grouped into categories which are useful for assessing the production of aromatics and butadiene, as well as the obvious identification of caustic soda and hydrogen. "Other By-products" are used as fuel.

3.1.2

Ì

.2 The Production of Dioctyl Phthalate from Propylene in West Pakistan

The alternative use of propylene to produce dioctyl phthalate was examined in the Interim Report, where it was concluded that, once a reasonable capacity for acetylene production is established in East Pakistan, there is a considerable saving to be gained in foreign exchange by producing dioctyl phthalate via the acetylene route in that wing. This conclusion was based on the comparison of schemes producing 16,000 MTPA of dioctyl phthalate completely from naphtha or completely irom natural gas.

The total production costs showed a very slight advantage to the propylene route. This difference, however, was within the limits of accuracy of the underlying estimates and assumptions.

| HUMPHREYS & GLAS | GOW LTD. | Volume II 3-3 |
|------------------------------|---------------------------------------|------------------|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser | C.1669 |

& Petrochemical Industries - Final Report

July 1970

Furthermore, it should be noted that in these comparative schemes the scale of naphtha cracking was very much larger than that for Phase II of the Fauji project, and thus yielded cheaper ethylene and propylene, whereas the scale of acetylene production was of the same magnitude envisaged for Phase II of the Ashuganj project.

Further consideration has been given to the conditions now applicable at the Fauji site. In view of the restricted ethylene capacity up to 1980, and the economic preference for maximising ethylene production for the manufacture of polyethylene, the manufacture of dioctyl phthalate from propylene would inevitably mean splitting the total Pakistan production between the two wings. The reduced scale of production would result in higher production costs at each petrochemical complex. Even if it were acceptable that no dodecyl benzene at all was produced, the total quantity of propylene available, including the surplus indicated in the Block Diagram II.3.1, would only be sufficient to produce 15,000 MTPA of dioctyl phthalate. Since the surplus of 1,300 MTPA of propylene mentioned above is too small to justify the production of dioctyl phthalate, some reduction in the 10,000 MTPA of dodecyl benzene would be necessary. The DDB production rate is already well below the estimated market demands for 1980, and not only would foreign exchange expenditure increase because of the production of dioctyl phthalate from propylene as indicated earlier, but it would also increase because of the import of dodecyl benzene, or a substitute detergent. These undesirable results can be avoided by the manufacture of the total Pakistan requirements for DOP from acetylene in East Pakistan, where such manufacture would contribute appreciably to the development of the petrochemicals industry in that wing.

A final aspect for consideration arises from the development of aromatics production in West Pakistan. The production of phthalic anhydride in this wing would offer some savings in transport costs if the manufacture of dioctyl phthalate were at Korangi. The foreign exchange savings resulting from production of DOP in Pakistan will not, however, be affected by the selection of East or West Pakistan as the plant location. The transport of phthalic anhydride from West to East Pakistan, and the transport of dioctyl phthalate back to West Pakistan would result in slightly increased cost to the consumer in West Pakistan.

3.1.3 Consideration of Other End-Products

1

ŧ

Apart from the end-products discussed above, the other end-; products listed in Table II.2.1 were considered.

Some of these, e.g. Urea-Formaldehyde resins, would obviously be produced at the Ashuganj site in East Pakistan because of the availability of urea and of formaldehyde from methanol. High density polyethylene is considered to be uneconomic at the scale indicated by the 1980 market demands.

The acetylene route is recommended for production of vinyl acetate and acetic acid. Therefore, these products are not recommended as outlets for ethylene in Phase II of the Fauji project. These products and polyvinylacetate are discussed further in Section 4.1.



HUMPHREYS & GLASOON LIMITED

VOLUME II

| | IOW LTD. | Volume II 3-5 |
|--------------|---|-------------------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | Ju ly 1970 |

3.2 <u>Recommended Development and Financial Evaluation for Fauji Foundation</u>, <u>West Pakistan</u>

3.2.1 Introduction

4 #

1

The recommended development for Phase II of the Fauji Project is shown in the Block Diagram II.3.1 and its evolution has been discussed in Section 3.1

Before considering the development scheme as a whole, the individual process units are described below; capital and operating costs, and the foreign exchange analysis are dealt with in a subsequent section. This part of Volume II concludes with the financial analysis of the scheme as a whole including the year-by-year cash flows and the D.C.F. rate of return on the investment.

3.2.2 Individual Process Units

3.2.2.1 Dodecyl Benzene Plant

Process Description and Flow Chart

A simplified flow chart for the production of dodecyl bensene is presented in Table II.3.2. The simplified flow charts omit such items as pumps, blowers and auxiliary systems (e.g. hot oil system).

The first part of the process may be considered as the preparation of propylene tetramer. Propylene from the ethylene plant is washed, vapourised, and polymerised over solid, phosphoric acid-based catalyst maintained in separate beds in the reactor, suitable propane quench points being provided between the beds for temperature control. The reactor effluent is depropanised, the liquid overheads being recycled (because of the unreacted propylene content) or used for The gaseous overheads is withdrawn as fuel gas. quench. The depropaniser bottoms pass to a recycle column, from which light polymer is withdrawn overhead; some of this is recycled to achieve further polymerisation to the tetramer. The bottoms from this column are purified in the tetramer column, from which heavy polymer is also withdrawn.

In the second part of the process, propylene tetramer is dried, and then reacted with benzene in the presence of hydrofluoric acid catalyst. The use of aluminium offloride gives lower yields, higher corrosion rates, and handling difficulties; it is not recommended. Supporte acid catalyst has been proved acceptable commercially, but gives relatively high utility and catalyst consumptions, and somewhat lower yields. More commercial experience has been gained with the hydrofluoric acid process.









۰.

SECTION 4



f'

SECTION 5

VOLUME I

| HUNPHREYS & GLAS | GOW LTD. | Volume II 3-7 |
|------------------|--|------------------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Inustries - Final Report | July 1970 |

A recirculating reactor system is employed. Acid is removed from the net reactor effluent in a settler. Most of this is returned to the reactor, while a small proportion passes through the hydrofluoric acid regenerator. The hydrocarbon layer from the settler passes to a stripper, the overheads from which combine with the regenerator overheads, and after condensation are recycled to the reactor and used as regenerator reflux. The regenerator bottoms, tar and polymer, pass to a neutralisation pit. The stripper bottoms pass to the benzene column. Benzene is taken overhead and recycled to the reactor. Only a very small portion of this recycle would require neutralisation (to prevent the build up of impurities) if high quality benzene feed is used; this is not shown on the flow chart. The bottoms from the benzene column are passed through a bauxite tower to remove combined fluorides, and light and heavy alkylates are separated in the Light Alkylate and Finishing Columns respectively.

The dodecylbenzene is a water-white liquid having a boiling range of 280°C, to 320°C.

Source of Know-How and Licence

The process scheme and operating data given subsequently have been based on the Universal Oil Products' process. An alternative licensor is Chevron.

Operating Data

Table 11.3.2 below lists the major consumptions of raw materials, catalysts/chemicals and utilities, and direct labour requirements.

TABLE II.3.2

| Operating | Data - | D ode cyl | Benzene | (Basis | 1 MT DDB |) |
|-----------|--------|------------------|---------|--------|----------|---|
| | | | | | | |

| Propylene (as 100%) MT | 1.380 |
|-------------------------------|-------|
| Benzene (nitration grade) MT | 0.472 |
| Net Byproducts, MT | 0.850 |
| Catalyst/Chemicals, \$ | 4 |
| Power, kwh | 232 |
| Steam (medium pressure) MT | 7.4 |
| Cooling Water, M ² | 31 |
| Labour Requirement | 18 |

Area of Battery Limits

The approximate area within the plant battery limits is 13,000 M².

| | | Volume II | Volume II | |
|--------------|---------------------------------------|-----------|-----------|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | | |

3.2.2.2 Polyethylene Plant

Process Description, Source of Know-How and Licence

The extension of the polyethylene plant from 15,000 MTPA to 38,600 MTPA (in 300 operating days per year) will require the addition of a second stream. It is highly probable that, in the interest of economy of spares, integration of the two streams, and utilisation of operating know-how gained on the first stream, the second stream will also utilise the I.C.I. process.

Operating Data

Table II.3.3 below lists the major consumptions of raw materials, catalysts/chemicals, utilities, and direct labour requirements.

| TABLE | TT | 3.7 | 5 |
|-------|----|-----|---|
| | ** | • | / |

Operating Data - Polyethylene (Basis 1 MT)

| Ethylene (as 100%) MT Catalysts/Chemicals, \$ | 1.070 15 |
|--|-------------|
| Power, kwh | 2000 |
| Steam (300 psig, saturated) MT | 1.5 |
| Steam (100 psig, saturated) MT | 0.6 |
| Cooling Water, (30°C), M ² | 200 |
| Condensate, M ² | 0.2 |
| Nitrogen, Nm ² | 7 |
| Labour Requirements | 50 |

Area of Battery Limits

The approximate area within the plant battery limits is 13,000 M². This is based on the overall plot plant for Phase I of the Fauji complex obtained from Simon Carves Ltd., showing the plot area for the 15,000 MTPA plant.

3.2.2.3 Ethylene Plant

The operating data for ethylene production is required for evaluating the variable cost of production in Section 3.2.6 'Foreign Exchange Analysis) and the average cost of production in Section 3.2.7 (Cash Flow). The major consumptions of raw materials. catalysts and chemicals, utilities, and incremental labour requirements for expansion to 60,000 MTPA capacity are given below.

| WHITHREYS & GLASS | IOW LTD. | | Volume II 3-9 |
|-------------------|---|--------------------------|-------------------|
| JNIDO Vienna | Pre-Investment Studies for Fertiliser | | C.1669 |
| or Pakistan | & Petrochemical Industries - Final Repor | •t | July 19 70 |
| | TABLE II.3.4 | | |
| | Operating Data - Ethylene (Basis | 1 MT) | |
| | Naphtha, MT | 3.40 | |
| | Catalyst/Chemicals, \$ | 0.4 | |
| | By-Products | | |
| | Pyrolyss gasoline (hydro-treated), MT | 0.70 | |
| | Fuel Oil, MT | 0.15 | |
| | Light Gases, MT | 0.51 | |
| | Propylene, MT Power, kwh | 0,45 · · · 300 | |
| | Steam (high pressure), MT | 0.23 | |
| | Cooling Water, M ³ | 417 | |
| | Increase in Labour Requirements (30.000 to 60.000 MTPA) | 5 | |

It should be noted that the ethylene plant cracked gas and propylene compressors are driven by steam turbines. Electric motor drives are retained for the ethylene compressors.

3.2.3 Complete Development Scheme

3.2.3.1 Utilities and Offsites Installations

The utilities plants and any new off-site facilities needed for Phase II at the site must be an extension to those already provided by Fauji for Phase I of their project. The detailed provisions being made in Phase I are not available. We have therefore provided a summary of the utility consumptions demanded by the Phase II project outlined above. In carrying out subsequent cost estimating, unit costs for utilities have been used. These data were supplied by Fauji Foundation (via Chemical Consultants Ltd.)

With the exception of tank storage for raw materials and end-products, the same arguments apply to offsite installations. Facilities such as administrative office blocks, canteen and medical services, and workshop facilities, etc. are expected to be adequate for Phase II.

3.2.3.2 Utilities Consumption

Table II.3.5 below summarises the major utilities consumption resulting from the recommended development for Phase II of the Fauji project.

| 1 | | |
|------------------------------|--|---|
| | IOW LTD. | Volume II 3-10 |
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 July 1970 |
| | TABLE 11.3.5 Summary of Total Utility Requirements f | or Phase II |
| | | |
| • | Power, kn 89 | 60 |
| | Steem, MT/Hr 20 | •0 |
| • | Cooling Water, M2/Hr 27 | 00 |
| | Condensate/Demin Water (for process) 0. M ³ /Hr | 75 |
| 3.1 | .3.3 Management, Personnel and Labour | |
| | estimated as 68 men. This covers the requ 4-shift system of operation and the require It does not include: | irements for a ments for day men. |
| • | (a) Maintenance personnel (b) Plant Manager, or partial sup levels | ervision at higher |
| | (c) Administrative and "overhead" with the activities of these | staff associated plants. |
| | As far as assessing the costs of la management are concerned, categories (a) an by maintenance charges and overheads, categ for in the rate of 9000 Rs/year used for ea operating staff. This figure is a weighte which allows for management and supervision in a single organisation the supervisor : s constant. | bour, personnel and d (c) are covered ory (b) is allowed ch member of direct d cost of labour and assumes that, ubordinate ratio is |
| | The polyethylens plant extension, a the Fhase I installation, may fall under th one plant manager. The dodecyl bensene pl plant manager, however. | lthough larger than e control of the ant will require a |
| 3.2 | .3.4 Plot Plan of the Recommended Development | |
| | Drawing No. C.1669-L51-4 gives the the Phase II expansion. This is based on t plan for the site. The expansion thus for | plot plan showing he most recent plot ms a logically in- |
| | tegrated part of the whole complex. The disposal of effluents has been | viewed in a similar |
| | way to the provision of utilities installat pocted that the Phase I Effluent Disposal S able of handling the additional load. | ions, and it is ex- ystem will be cap- |
| HUNDHREYS & GLASGOW | LTD. | | 3-1 |
|--------------------------------|--|--|--|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertili & Petrochemical Industries - Final R | ser eport | C.1669 July 1970 |
| 3.2.4 <u>Capita</u> 3.2.4.1 | L Costs L Dodecyl Benzene Plant | | |
| <u> </u> | TABLE 11.3.6 | | |
| | Capital Cost of Dodecyl | Bensene Pla | nt |
| | | Rupees | <u>x 10</u> 6 |
| | | Foreign Exchange | Local Current |
| | 1. Equipment, spares, procure- ment, freight and insurance. tools and plant, insurance | 14.55 | - |
| | 2. Engineering, expenses and overheads, licence, site supervision and commis- sioning. | 4.72 | - |
| | 3. Construction labour, civil work, local tools and plant, local insurance, inland freight and dock clearance, local spares, local expenses | - | 5.48 |
| | 4. Duty on equipment and spares | • | 6.10 |
| | TOTALS | 19.27 | 11.58 |
| | Start-up Charges and Training Interest on Foreign Exchange during Construction | 1.54 2.12 | 1.54 |
| | | 22.93 | 13.12 |
| | TOTAL COST | × | .05 |
| | ESTIMATED WORKING CAPITAL (Max.) | 4 | .00 |
| | The above capital cost estimate cove are normally within the plant batter penses required to bring the plant i | rs all fixtury limit, and nto operatio | ures which all ex- m. |
| | Included: - Raw materials and product sufficient for normal operation. Of minor office buildings and workshops | t storage an ivil work, c , fire equip | nd handlin control ro pment, tel |
| | phones etc. within the plant boundar <u>Excluded</u> : - Major offsite storage far piping, and services distribution of site development and land | y, as necess wilities, ci utside plant | vil work, boundary |
| | Interest during construction on the omitted because details of the method | rupee elemen d of finance | nt is are not |

Í

4

| | ow LTD. | | | Volume 1 3-1 |
|--------------|------------|---|--------------------|--------------------------|
| UNIDO Vienna | Pre | -Investment Studies for Fertil | iser | C.166 |
| for Pakistan | & Pet | rochemical Industries - Final | Report | July 197 |
| | | | | |
| 3.2 | .4.2 Polye | thylene Plant | | |
| | | TABLE II.3.7 | | |
| | | Capital Cost of Polyethy | lana Plant | |
| | | | | L L |
| | | | Nupees | <u>x 10⁶</u> |
| | | : | Foreign Schenge | Local <u>Currency</u> |
| | 1. | Muipment, spares, procure- ment, freight and insur- ance, tools and plant, insurance | 25.40 | - |
| | 2. | Engineering, expenses and overheads, licence, site supervision and commis- sioning | 19.47 | • |
| | 3. | Construction labour, civil work, local tools and plant, local insurance, inland freight and dock clearance, local spares, | - | 9.39 |
| | 4. | Duty on equipment and spares | • | 10.55 |
| | TOTALS | | 44.87 | 19.94 |
| | Start- | up Charges and Training | 3.24 | 3.24 |
| | Intere | st on Foreign Exchange during | 4.94 | - |
| | | | 53.05 | 23.18 |
| | TOTAL | CORT | 76. | |
| | | THE WORKING CAPTERI (MAX) | 11. | 64 |

| HUMPHREYS & GLASS | IOW LTD. | Volume II 3-13 |
|-------------------|---|-------------------|
| UNIDO Vienna | Pre-Investment Studies for Pertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

3.2.4.3 Expansion to Phase II Ethylene Plant

Chemical Consultants Ltd. have provided the foreign exchange cost which Fauji Foundation will incur in expanding the capacity of the ethylene plant from 30,000 to 60,000 MTPA. The foreign exchange cost of expanding the utilities installation for this increased ethylene capacity has also been provided. These costs have been used in the Foreign Exchange Analysis in Section 3.2.6, and are tabulated below.

| TABLE II.3.8 | | |
|--|--------------------------|---|
| | Rupees x 10 ⁶ | |
| | Foreign Enchange | Local <u>Currency</u> inclu- ding duty |
| Ethylene Plant Expansion | 7.14 | |
| Utilities Installation Expension | 11.90 | |
| Allowance for Breetion, and Commissioning | 2.62 | |
| | 21.66 | 13.12 |
| during Construction | 2.38 | |
| TOTAL | 84.04 | 13.12 |
| TOTAL COST | 31 | . 16 |

The local currency capital cost given in the above Table is based on the cost analysis of typical petrochemical plants.

| | | IOW LTD. | | Volume II 3-14 |
|-------------|------------------------------|--|---|---|
| | UNIDO Vienna for Pakistan | Pre-Investment Studies for Perti & Petrochemical Industries - Pinal | liser Report | C.1669 July 1970 |
| | | | | |
| | 3.2 | 2.4.4 Summary of Capital Costs | | |
| ľ | | TANKE II 3.9 Summerry of Capital Casta of 1 | Revid Russe | |
| • | | | Rex 1 | .0 ⁶ |
| l l | | | Poreisn Exchange | Local Currency (including duty) |
| l l | | Estimated cost of Fauji Phase I including pre- investment for expansion, bags and pipe plants, utilities, start- up costs, and interest on foreign exchange (but not local currency): | 216.33 | 128.25 |
| I | | TOTAL COST: ESTIMATED WORKING CAPITAL: | , | 44.58 18.00 |
| I | | Dodecyl Benzene Plant Second Polyethylene Plant TOTALS FOR NEW PLANTS Start-up charges and training Interest during construction | 19.27 44.87 64.14 4.78 6.64 | 11.58 19.94 51.52 4.78 (b) - (4 ,00) |
| - | | | 75.56(a | 36.3 0(40.3 |
| ÷ | | TOTAL COST OF NEW PLANTS: Ethylene Plant Extension Interest during construction | 21.66 2.24 | 11.86 |
| | | TOTAL COST OF ETHYLENE EXTENSION | | <u>15.12</u> 37.02 |
| | | New Plants plus ethylene extension | 00 16 | to ha |
| | | TOTAL COST OF EXPANSION: ESTIMATED WORKING CAPITAL: | <u> </u> | 48.88 |
| • 4 1 | | The costs given above for Phase I a Fauji, and were provided through Ch | ure those e: Hencon. | stimated by |
| • | | The capital costs for the new plant flow calculation, in Section 3.2.7. costs of ethylene include an elemen costs of the complete ethylene plan amounting to $\mathbb{D}_{0}^{(k)}$ is added to the lo | a) a are used 6, in which it to cover it and offs: cal current | for the cash h the average the capital ites. 'Interest |

| HUMPHREYS & GLASG | HUMPHREYS & GLASGOW LTD. | | | |
|-----------------------|---|-----------|--|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | | |
| for P akis tan | & Petrochemical Industries - Final Report | July 1970 | | |

3.2.5 Unit Costs and Prices

Table II.3.10 below lists the unit costs of the various raw materials and utilities used in the subsequent financial analyses. It also gives the credit values for the various byproducts and the expected selling price of the end-products (including ethylene sold to Valika). The costs, values, and prices are subdivided as shown, for the purpose of specific financial analyses.

In the table, the "total" prices for polyethylene and dodecylbenzene are those proposed by Fauji (before sales tax), as quoted to us by Chemcon. The foreign exchange prices for these end-products are the expected C & F prices of the replaced imports, as in our Interim Report, and are based on current C & F prices. The foreign exchange prices shown in brackets are an estimate of the C & F price which would be competitive, under cash cum bonus, with the proposed total price.

The "total" price of ethylene sold is H & G's own estimate, based on approximately 50% mark-up of our estimate of the production cost of ethylene as stated in 3.2.7.2 below, to cover liquefaction, transport, and profit.

| | TABLE 11.3.10 | | |
|--|------------------------------------|--|-----------------------------|
| <u>Un</u> | it Costs and Pr | ices | |
| (\$ |) \$ Foreign <u>Exchange</u> | Rs Local Currency (exclud- <u>ing duty</u>) | Rs <u>Total</u> |
| End Products Polyethylene, MT (244 Dodecyl Benzene MT (203 Ethylene MT | 8) 270 2) 30 0 | - | 2960 2400 1000 |
| By Products Pyrolysis Gasoline (Hydro- treated | 14.2) MT | 52 | - |
| Ethylene Plant) MT Byproduct ex DDB plant MT Haw Materials Light Naphtha MT | - 14.3 | 21 70 14 | - 86 |
| Benzene (local manufacture) Utilities Electricity (local purchase |) MT - | 700 | 700 |
| Steam, MT Cooling Water, M ³ Raw Water M ³ Demineralised Water M ³ | - - - 0.05 | 5.7 0.06 0.44 0.61 | 15.0 0.06 0.44 2.0 |
| Packages Sacks (per MT Polyethylene) Steel Drums (per MT DDB) |) - | 8.0 60.0 | 8.0 60.0 |

| | | Volume IT |
|---------------|--|---|
| HUMPHREYS & G | ASGOW LTD. | 3-16 |
| UNTDO Vienna | Pre-Investment Studies for F | Vertiliser C.1669 |
| for Pakistan | & Petrochemical Industries - F | inal Report July 1970 |
| 3.2.6 | Foreign Exchange Analysis | |
| | 3.2.6.1 <u>General</u> | |
| | of Phase I of Fauji is shown i It is assumed that in Phase I, produced for Valika. It show Phase I and Phase II, operatin with the production of 0.45 MI the surplus propylene (burnt a MTPA respectively. It is, th remains sufficiently in balance analysis. | In the Block Diagram II.3.1. 5,300 MTPA of ethylene are ald also be noted that for both ag at maximum severity cracking of propylene per MT of ethylene, as fuel) is 1700 MTPA and 1300 berefore, assumed that propylene be for the purposes of this |
| | The incremental product and dodecyl bensene is summari | tion of ethylene, polyethylene sed below: |
| | TABLE 1 | 1.3.11 |
| | Incremental Pro | oduction Rates |
| | | MTPA |
| 1 | Ethylene | 30,400 |
| l | Polyethylene | 23,600 |
| | Dodecyl Bensene | 10,000 |
| | The basis of the foreign evaluation of the foreign exch cost (excluding duty) for hence the statement of the cost foreign exchange. It is appl ties in the above Table 1 | Ign exchange analysis is the hange saving and the local curren this incremental production, and st in local currency of saving lied specifically to the quanti- to full rate production |
| J | 3 6 2 Venichle Costs of Destuction | A THE TANK IS A MARKATANI |
| 2 | J.C.O.C VERIEDIC COSTS OI Production | |
| • | From the operating dat the unit operating costs given | n in Section 3.2.5, the variable |
| | costs of production for ethyle benzene is cslculated. In or | ene, polyethylene, and dodecyl rder to make the figures independe |
| | of any judgement on the relat: | ive unit values of ethylene and p |
| 1 | pyrene, the variable cost of (| cost of these intermediates is |

| | 50W LTD. | | | 107 - 1 0 - 10 - 10 - 10 - 10 - 10 - 10 - 10 | Volume II 3-17 |
|------------------------------|----------------------|---|----------------------------|---|---------------------------------|
| UNIDO Vienna for Pakistan | Pre-In & Petroc | vestment Studies for hemical Industries - | Fertiliser Final Repo | , ort | C.1669 July 1970 |
| | | | | | |
| | | TABLE | 11.3.12 | | |
| | | Variable Pr | oduction Co | <u>ets</u> | |
| | | | | Foreign Exchange | Local Cost Excluding Duty |
| | | Besis | | (#) | (Rs.) |
| | Ethylene + Propyl | 1 MT) ene 0.45 MT) | | 18 O | 5.k % |
| | Polyethy | lene 1 MT (excl. et) | vlene cost |) 15.0 | 27.7 |
| | Dodecyl | Benzene 1 MT (excl. | propylene | 4.0 | 394 |
| | Notes on | Table II.3.12 | | | |
| | 1. | The local costs in t packages. | the above Tr | ble inclu | de the cost of |
| | 2. | Summation of the var total variable costs calculated, | iable costs of the com | listed en plex exten | nables the nsion to be |
| 3.2. | 6.3 Plxed Co | ets of Production | | | |
| 20-0 | Maintena | noe | | | |
| | | | | | |
| | percenta | the basis for calcul se of the total fixe | ating the m d capital (| aintenance excluding | oharges as a interest |
| | during of | onstruction, site de | velopment, | etc.) is: | |
| | 1 | Poreign Exchange | 1.8# | | |
| | _ | | | | |

This basis makes due allowance for materials and labour and is applicable to Pakistan conditions.

For the purpose of this calculation, however, it was more convenient to base the maintenance on the foreign exchange investment and the percentages used were

.

Ľ

| HUMPHREYS & GLAS | GOW LTD. | Volum |
|------------------|--|---|
| UNIDO Vienna | Pre-Investment Studies for Pertiliser | C.: |
| TOP FARISTAN | a Petrochemical Industries - Final Report | July |
| | Foreign Exchange element 2.9% of for of maintenance costs capital cost | vign exchang |
| | Local Cost element of 2.5% of for maintenance costs capital cos | eign exchang t |
| | From the total foreign exchange invest β 18.0 x 10 ⁶ (see Section 3.2.4), the maintena annum is: | ment of noe charge p |
| | Foreign Exchange \$ 522,000 | |
| | Local Cost (excluding Rs. 2,147,0 duty) | 00 |
| | Labour and Supervision | |
| | expansion, the additional polyethylene plant, a bensene plant are 73 men (see Section 3.2.2). cost of 9,000 Rs. per man per year, which allow vision, the labour charges are: | and the dode At an aver We for super |
| | | |
| | Local Cost Re. 657,000 | |
| | Local Cost Re. 657,000 | |
| | Local Cost Rs. 657,000 <u>Overheads</u> These, which include selling and market are estimated at 3% of the total sales revenue. culated using the data from Table II.3.10, thus | ing expense This is contract |
| | Local Cost Rs. 657,000 <u>Overheads</u> These, which include selling and market are estimated at 3% of the total sales revenue. culated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 | ing expense This is c Rs. x 10 ⁶ |
| | Local Cost Rs. 657,000 Overheads These, which include selling and market are estimated at 3% of the total sales revenue. oulated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 10,000 MTPA DDB 24.0 | ing expenses This is co Re. x 10 ⁶ Re. x 10 ⁶ |
| | Local Cost Rs. 657,000 <u>Overheads</u> These, which include selling and market are estimated at 3% of the total sales revenue. culated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 10,000 MTPA DDB 24.0 5,200 MTPA Ethylene 5.2 | ing expense This is co Re. x 10 ⁶ Re. x 10 ⁶ Re. x 10 ⁶ |
| | Local Cost Rs. 657,000 <u>Overheeds</u> These, which include selling and market are estimated at 3% of the btal sales revenue. culated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 10,000 MTPA DDB 24.0 3,200 MTPA Ethylene 5.2 99.2 | ing expense This is of Rs. x 10 ⁶ Rs. x 10 ⁶ Rs. x 10 ⁶ Rs. x 10 ⁶ |
| | Local Cost Rs. 657,000 <u>Overheeds</u> These, which include selling and market are estimated at 3% of the total sales revenue. culated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 10,000 MTPA DDB 24.0 5,200 MTPA Ethylene 5.2 99.2 The overheeds are thus Rs. 2,976,000 (1 | ing expenses This is of Rs. x 10 ⁶ Rs. x 10 ⁶ Rs. x 10 ⁶ Rs. x 10 ⁶ Rs. x 10 ⁶ |
| | Local Cost Rs. 657,000 <u>Overheads</u> These, which include selling and market are estimated at 3% of the total sales revenue, oulated using the data from Table II.3.10, thus <u>Sales</u> 23,600 MTPA Polyethylene 70.0 10,000 MTPA DDB 24.0 3,200 MTPA Ethylene 5.2 99.2 The overheads are thus Rs. 2,976,000 (1 <u>Interest on Working Capital</u> | ing expense This is c This is c Rs. $x 10^6$ Rs. $x 10^6$ Rs. $x 10^6$ Rs. $x 10^6$ Rs. $x 10^6$ coal cost or |

| U -e | HUNPHREYS & GLA | SOW LTD. | Volume TI 5-19 |
|---------|-----------------|---|--------------------------|
| | UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

The interest on working capital is thus Rs. 1,438,000 for 100% production capacity.

The fixed costs of production are summarised below-

TABLE 11.3.13

Fixed Production Costs

| | Poreign Rebange | Local Cost Excluding Duty |
|--------------------------------|---|---------------------------------|
| | # x 10 ³ /yr | Re. x 10 ³ /yr |
| Maintenance | 522 | 2147 |
| Lebour and Supervision | NI 1 | 657 |
| Overheads and Selling Expenses | M11 | 2976 |
| Interest on Working Capital | N11 | 1488 |
| | | - |
| TUTAL | 522 | 7 26 8 |
| | and the second se | |

3.2.6.4 Poreign Exchange Saving and Ruppe Cost Comparison

The following Table gives the comparison and is compiled using the data given in the previous sections. The Rupee element of the investment excludes duty on equipment, but includes start-up expenses. The use of the average interest rate ensures that the early years are not penalised, nor the results for the later years over optimistic.

| HUMPHREYS & GLASCO | Volume II 3-20 | |
|--------------------|---|-----------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

I

I

Ĩ

- States

I

Ĩ

I

Ţ

Ξ.

TABLE 11.3.14

Foreign Exchange Analysis (Pauji - Phase II)

| | Poreign Exchange | Local Cost Excluding Duty |
|--|------------------------|---------------------------------|
| | # x 10 ³ /y | r Re. x 10 /yr |
| Variable Production Costs | | |
| P.E. 23,600 MTPA | 354 | 4810 |
| DDB 10,000 MTPA | 40 | 3940 |
| Ethylene 30,400 MTPA | 1182 | 435 |
| Fixed Production Costs | 522 | 7268 |
| TOTAL PRODUCTION COSTS | 2096 | 16453 |
| Foreign Exchange Value of Products | | |
| P.E. 23,600 MTPA @ 270 S/MT | 6370 | |
| DDB 10,000 MTPA @ 300 \$/MT | 3000 | |
| Ethylene 5,200 MTPA | N11 | |
| TOTAL F.E. VALUE OF PRODUCTS | 9370 | |
| Average Interest Over First 10 Years at 35 p.a. (# 4% on Total Foreign Exchange Loan of \$ 20.92 x 10 ⁵) | 837 | |
| Annual Repayments | 2092 | |
| TOTAL INTEREST AND REPAYMENTS | 2989 · · | |
| Foreign Exchange Savings | 4343 | |
| COST OF F.E. SAVING RE/S EXCLUDING DEPRECIATION ON RUPEE ELEMENT | 3. | 79 |
| Depreciation @ 10% on Rupee Element | | 2584 |
| Rupee Cost Including Depreciation | | 19037 |
| COST OF F.E. SAVING RE/S INCLUDING DEFRECIATION | 4.3 | 6 |

| | HUMPHREYS & GLASGON | N LTD. | | Volume II 3 - 21 |
|---|---------------------|--|--|--|
| T | UNIDO Vienna | Pre-Investment Studies | for Fertiliser | C.1669 |
| _ | for Pakistan | & Petrochemical Industrie | as - Final Repor | t July 1970 |
| l | 3.2.6.5 | Foreign Exchange Analysis | for Each Produc | t |
| | | The variable and f propylene, polyethylene, a calculated, on exactly, th sections. The cost of 1 MT (both operating and finance between the ethylene expa before, the duty and bonus the costs of plant, materi costs, as before, cover re the foreign exchange eleme only on the local currency interest on working capita | 'ixed production and dodecylbenze as a me basis as T of propylene of ethylene, an dial) have been ansion and the p elements have als and utility payments and av out of fixed cap r element of fixed. | a costs of ethylene, one, have been in the preceding has been taken to id the fixed costs suitably distributed process units. As been omitted from services. Financial erage interest on ital, depreciation ed capital, and |
| I | | The production cos figures so calculated are: | ts and foreign | exchange savings |
| T | | | \$ | Rs |
| I | | | reign Exchange | Local Currency |
| T | | per (1 MI eunylene + 0.45 | MI propylene): | |
| I | | Variable Fixed | 38.9 28.7 | 14.3 96.7 |
| ĭ | | Production cost | 67.6 | 111.0 |
| 1 | | ethylene, per MT: | | |
| 1 | | Fixed | 23.4 | <u>78.9</u> |
| | | Production cost | 55.2 | 90.6 |
| • | | Variable | 15.0 | 5 8 |
| - | | Fixed | 11.7 | 34.5 |
| • | | Production cost | 27.0 | 45.3 |
| | | Ethylene 1.07 MT | 59.0 | 96.7 |
| Ì | | Other variable Fixed Production cost | 15.0 <u>76.5</u> 150.5 | 204 257 557-7 |
| | | Cost of import (C&F) Foreign exchange | 270 | <i></i> |
| • | | saving | 119.5 | |
| | | Dodecylbenzene, per MT: | 9.) = RE 4.7/\$ | |
| | | Propylene 1.38 MT Other variable | 38. 0 4.0 | 62.5 394 |
| | | Fixed Production cost Cost of import | 1 77.4 1 19.4 | 260 716.5 |
| | | (C&F) Foreign exchange | 300 | |
| | | saving | 180.6 | |

Foreign exchange saving 180.6 Ratio Rs 716.5/\$ 180.6 ± Rs 4.0/\$

| | HUMPHREYS & GLASG | OW LTD. | Volume II 3 - 22 |
|---|-------------------|---|---------------------|
| Į | UNIDG Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| 1 | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

3.2.7 Cash Flow and D.C.F. Rate of Return

3.2.7.1 General

The sales revenue is based on the pattern of market build-up stated in the Interim Report. Production by the individual plants 's only limited to 30% and 90% in the first two years of operation as existing plants and utilities will be well established. The lower figure of the two (market requirement or production capacity) establishes the output (and sales rate) for any particular year up to 1980, after which the full production rate can be maintained and sold. For the purposes of this analysis it was decided to use the average, and not the marginal, cost of ethylene and propylene production in establishing the production costs of the polyethylene and dodecyl benzene (see Section 3.2.7.2 below). Total costs of the manufacturer are used through this analysis so that the D.C.F. rate of return can be determined from the viewpoint of the Pakistan investor.

3.2.7.2 Ethylene and Propylene Costs

In Phase II, the Fauji plant is planned to produce 60,000 MTPA ethylene with a propylene to ethylene ratio of 0.45 under severe cracking conditions. For the purposes of analysing Phase II of the project, ethylene and propylene are taken as feeds to the new plants proposed. Propylene is regarded as a less valuable product than ethylene and, even after Phase II, there will be surplus production from the Fauji cracker. The cost of propylene per MT has been taken as half that of ethylene. Both ethylene and propylene costs have been averaged over total production; the Phase II plants do not therefore tenefit from ethylene and propylene at marginal costs of production.

The estimated total costs of ethylene and propylene as feeds to the Phase II units are:

| Ethylene | Rs. | 635 | /MT |
|-----------|-----|-----|-----|
| Propylene | Rs. | 317 | /MT |

3.2.7.3 Variable Cost of Polyethylene and Dodecyl Benzene Production

Using the average production costs of ethylene and propylene from the previous section, the operating data given in Section 3.2.2, and the total unit cost of raw materials and utilities from Section 3.2.5, the variable costs of polyethylene and dodecyl benzene were calculated. These are summarised below:

| | | WY LTD. | Volume II 3 - 23 |
|--------|------------------------------|--|--|
| | UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 Ju ly 19 70 |
| J | | | |
| J | | Polyethylene | Rs. 1080 /MT |
| | | Dodecyl Benzene | Rs. 932 /MT |
| | | These costs include the cost of pa | ckages. |
| | 3.2.7 | .4 Fixed Costs of Production | |
| | | The fixed costs are estimated for a ject as a whole and are not subdivided betw In the cash flow table and D.C.F. calculat: of the full capacity ethylene plant on the | the extension pro- ween end-products. ion, the presence site is assumed. |
| 1 | | Maintenance | |
| | | The basis for the maintenance charge total fixed capital (excluding the duty pai This is consistent with the rates given in On this basis the figure of 4.27% includes duty element on the materials used in maint | ge is 4.27% of the id on equipment). Section 3.2.6.3. allowance for the tenance. |
| | | Labour and Supervision | |
| 1 | | The direct labour requirements for 68 men, charged at 9,000 Rs. per man per ye | the two plants are par. |
| 18 | | Overheads and Selling Expenses | |
| U 1 | | These are estimated to be 3% of the as in the Foreign Exchange Analysis. They "full-up" year as by definition they are co fixed. | total sales revenue, are based on the ensidered to be |
| n | | Foreign Exchange Repayments | |
| U P | | The repayments are made in equal su ten years of operation. The total foreign the polyethylene and dodecyl benzene plant | ms over the first exchange loan for |
| IJ | | Rs. 75.56 x 10^6 | |
| F | | It includes interest on the loan during the period and the foreign exchange capital cov | construction ering start-up |
| Γ | 3.2.7. | 5 Other Fixed Costs | |
| Γ | | Interest on Foreign Exchange Loan | |
| | | This is based on 8% of the outstand: cludes interest on the foreign exchange cap: cover start-up expenses. | ing loan. It in- ital borrowed to |

| HUMPHREYS & GLAS | IGOW LTD. | Volume II 3 - 24 |
|------------------|---|---|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C,1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |
| | Interest on Working Capital | |
| | As in the Foreign Exchange Analysis. 1.5% of the total sales revenue. The total applicable to the particular year is used. excluded. | this is based on sales revenue Ethylene is |
| 3. | 2.7.6 Cash Flows and D.C.F. Rate of Return | |
| | The cash inflows, cash outflows and t are set out in Table II.3.2.15 below. The r before tax and depreciation. | the net cash flows net cash flow is |
| | A total span of 15 years is covered; reference year in which it is assumed that al invested. In practice this is not the case, ance mentioned above for the interest during permits this simplification. 1976 is taken which the recommended development commences p sales. It is justified by the D.C.F. rate o | year 0 is the 1 the capital is but the allow- construction as the year in roduction and f return obtained |
| | The costs in this table are based on equipment, materials and labour. This is in accepted practice where price trends are not Labour is the only element of costs for which reliable trend prediction is available. How cost is a very small element in this table, a reductions of the work force might be made ov the plant by improved methods, to counteract and salaries. Therefore, allowance for risin not considered to be useful in the present c | 1970 costs of accordance with available. a reasonably ever, labour nd in practice er the life of rising wages ng labour cost wa ase. |
| | The proposed selling prices are those Fauji Foundation, as quoted to us by Chemical They are well below the current import prices bonus; the C & F prices which would be compe- proposed selling prices have been estimated an Table II 3.10. For explanation of capital con Section 3.2.4.4. | estimated by Consultants. on cash-cum- titive with the nd are shown in sts used, see |
| | The D.C.F. rate of return based on the ment in the fixed assets is calculated at | Rupee invest- |
| | | |
| | | |
| | | |

+

l

(.....

-

| | | · W. | | | | 2 - 25 |
|------------------------------|------------------------------|--|-------------------------|--------------------------|---|---------------------|
| UNIDO Vienna for Pakistan | 6 | Pre-In t Petro | nvestment chemical] | Studies fo Industries | or Fertiliser - Final Report | C.1669 July 1970 |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | 2 | | | 못봐 | 19135998 | |
| | 1 1 | 333 | REEM | | . S. SENSTR | |
| | | | *** W | **** | | |
| | | 777 | 732 | 222 | 13132435 | |
| | 1 2 | | | 3 3 박사 | 13 135 135 | |
| | 1 = | | 838 M | 3393 | | |
| | 1 9 | | 838 M | *** | | |
| | | * # 4 | ×83 6 | 2322 | - 2 2 2 2 3 3 3 3 3 | |
| | | 7 3 3 | | | | |
| | | | | 112 | | |
| _* | | 888 | 888 M | | 100 10 10 10 10 10 10 10 10 10 10 10 10 | |
| | | *#4 888 | 233 b | 2222 | 22 222 200 | |
| | | 773 | 732 5 | 7375 | | |
| | 1 | 1 1 1 1 1 1 1 1 1 1 | 339 P | 33 9 3 | 11 11 11 11 11 11 11 11 11 11 11 11 11 | |
| | E - | 333 | 444 M | 11.84 | | |
| 44 6 8 | E | | ***** | | 2.2 2.2 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 | |
| 5 | | | | 127 | 33 232 78 | |
| | £ • | 199° | *230 | | şa şəş əhərə | |
| | ¥ - | 188 | | - 5四 | 20 536000K | |
| | 6. | | - 3 - 10 | | • FUE AND | 2* |
| | Ĭ <u>−</u> <u>+</u> . | | | | | 19. 19. |
| | | | | | | |
| | | | | | | |
| | | G.I L | | | | 50 7 |
| | | | | 472.3 | | ¥ |

| HUMPHREYS & GLASG | HUMPHREYS & GLASGOW LTD. | | | | |
|-------------------|---|-----------|--|--|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | | | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | | | |

3.3 Conclusions

A NUMBER OF STREET

1. The extension of the ethylene plant capacity to 60,000 MTPA for Phase II of the Fauji Foundation complex makes an additional 30,400 MTPA of ethylene available. It is recommended that this is used to produce:

23,600 MTPA polyethylene

10,000 MTPA dodecyl benzene

5,300 MTPA ethylene (to Valika)

These products are the most beneficial from the national and commercial points of view. A scheme including these products has been developed and a start-up date of 1976 is proposed.

2. The Rupee cost of saving foreign exchange is estimated at Rs. 4.38/ U.S. \$, and is very favourable.

Ethylene sold to Valika is eventually transformed to polyethylene which saves further foreign exchange and improves this figure.

3. Taking ethylene as a feedstock from the extended cracker, the D.C.F. rate of return on new Rupee capital invested in the polyethylene and DDB plants has been calculated at 43.8% (before tax). This is a high rate of return on the Rupee capital; the figure would be much smaller if calculated on the total project cost because of the high proportion of foreign exchange cost in the units.

4. The substantial savings in foreign exchange and attractive D.C.F. rate of return indicate that a scheme of this type would still be viable at lower capacity. Alternatively the scheme could be brought into operation earlier than the assumed date of 1976. An earlier date may, however, be difficult to achieve for the second phase of this project. Fauji Foundation should attempt to bring their Phase II into effect as early as possible.

| HUMPHREYS & GLA | HUMPHREYS & GLASGOW LTD. | | | |
|-----------------|---|-----------|--|--|
| UNTDO Vjenna | Pre-Investment Studies for Fertiliser | C.1669 | | |
| For Pakistan | & Petrochemical Industries - Final Report | July 1970 | | |

SECTION 4

E.P.I.D.C., EAST PAKISTAN PETROCHEMICAL COMPLEX

4.1 Analysis of Development Schemes for E.P.I.D.C., East Pakistan

No information was available to H & G regarding the further development of Phase I of the E.P.I.D.C. scheme at Ashuganj. Accordingly, the developments for Phase II were logically derived from the planned production for Phase I, taking into account the recommendations made in Section 3 for Fauji Foundation.

From these recommendations, and the list of petrochemical products given in Table II.2.1, it can be seen that the following products can be considered for natural gas based manufacture at Ashuganj:-

> Dioctyl phthalate Polyvinyl acetate Vinyl acetate Acetic acid Urea/formaldehyde resins

As mentioned in Section 2.9, the production of dioctyl phthalate was recommended in our interim Report as being economically profitable from both the national and commercial points of view. Section 3.2 discusses in detail the reason for excluding dioctyl phthalate from the production scheme for the Fauji Foundation in West Pakistan; since the 1980 market demand for this product is considerable (16,000 MTPA for Pakistan as a whole), it is the major product for consideration in East Pakistan. Further consideration shows that the foreign exchange element of the total production cost of dioctyl phthalate is considerably less than the current C & F price in Pakistan. This product is therefore included in the recommended development scheme.

Section 2.11 discussed the requirements for polyvinyl acetate in detail and Section 2.12 explained that it is generally accepted that the additional polymerisation facilities would be provided by extensions to existing plants, situated at the point of usage. There are four existing and sanctioned plants in Pakistan (three in Karachi and one in Dacca) and the individual plant capacities would still be relatively small even after extension. The manufacture of polyvinyl acetate in one polymerisation plant at Ashuganj is not, therefore, recommended for Phase II.

The manufacture of vinyl acetate, on the other hand, is recommended for Phase II. The total market demand for 1980 is shown in Table 11.2.1 to be estimated at 8,420 MTPA for the whole of Pakistan and includes the supply of vinyl acetate to existing and sanctioned polymerisation units. The market demand is next to dioctyl phthalate in the list of products given earlier. Investigation showed that the foreign exchange element of the production cost, including both variable and fixed components, was less than the cheapest F.O.B. European source of supply (based on Dutch supply, early 1970 rates).

| HUMPHREYS & GLASO | Volume II 4-2 | |
|-------------------|---|-----------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

& Petrochemical Industries - Final Report

July 1970

When freight and insurance to Pakistan, and landing costs are added, the comparison is more favourable. This product is, therefore, included in our recommended development scheme.

The concomitant production of acetic acid, the precursor for vinvl acetate, makes it possible to satisfy the relatively small market demand for acetic acid, as a product in its own right. This demand is estimated to be 1,650 MTPA for 1980 for the whole of Pakistan. The total consumption of acetic acid for all purposes is 8,180 MTPA in 1980, and the foreign exchange element of the total production cost for this scale of production was calculated. This was found to be lower than the cheapest F.O.B. European source of supply (based on Belgian supply, early 1970), excluding freight, insurance and landing costs for Pakistan. In addition, the containers for this chemical are expensive and its internal manufacture, rather than importation, is justified.

The Block Diagram Figure II.4.1 summarises these conclusions and depicts our recommended development for this site. Table II.4.1 summarises the end products additional to Phase I.

TABLE II.4.1

END-PRODUCTS ADDITIONAL TO PHASE I

| | MTPA |
|-------------------|--------|
| Dioctyl Phthalate | 16,000 |
| Vinyl Acetate | 8,420 |
| Acetic Acid | 1,650 |

The manufacture of the remaining products in the list given above viz urea/formaldehyde resins, is not recommended. Although the raw materials for these resins are manufactured at Ashuganj (methanol being the source of formaldehyde), existing plants already take these materials from relatively large internal producers. This situation is considered to be analogous to that for the production of polyvinyl acetate, and it is considered that the additional market requirements will be provided by extensions to the existing and santioned plants, situated near the point of usage.

These resins vary considerably in their formulation, according to their end use, and many of them have a relatively short shelf-life. The additional 1980 market requirement of 7,200 MTPA for the whole of Pakistan is not very great, and it is our opinion that the present mode of manufacture by a number of small producers will continue.

| HUNPHREYS & GLASGOW LTD. | | Volume II 4 - 3 | |
|--------------------------|---|--------------------|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | |
| for P akista n | & Petrochemical Industries - Final Report | July 1970 | |

It seems appropriate at this point to amplify the reasons for H & G's recommendation of acetylene routes for the manufacture of dioctylphthalate, acetic acid and vinylacetate, contrary to the present world trend which is towards the use of ethylene for these chemicals.

The sanctioned Phase I of the EPIDC petrochemical complex, including 50,000 MTPA of PVC, is a starting point for this part (Volume II) of H & G's Study. However, H & G also recommended this route to PVC in our Interim Report, after examining a number of possible schemes. In effect, we found that the cost of such PVC production would be about the same <u>either</u> via acetylene <u>or</u> via ethylene supplied by a cracker of some 80,000 MTPA capacity, but the acetylene route (based on indigenous natural gas) would have a lower foreign exchange element of cost.

Further, comparison of the appropriate schemes in the Interim Report showed that, having decided to use the acetylene route to PVC it then becomes slightly cheaper to use the acetylene route also for dioctylphthalate, rather than either of the routes via ethylene or via propylene (see also Section . 3.2 of this Volume).

Acetaldehyde is an intermediate for 2-ethylhexanol and hence dioctylphthalate, from acetylene. Having decided to use this route, the cheapest way to meet the requirement for acetic acid is via acetaldehyde.

Acetic acid makes up the greater part of the raw material cost of vinyl acetate, whether ethylene or acetylene is used as the source of the vinyl group. Although there is a world trend towards the use of ethylene for new vinyl acetate production, this route is only attractive for fairly large scale production of vinyl acetate, based on cheap ethylene from a really large cracker. In Pakistan the estimate requirement is for only 8,420 MTPA of vinyl acetate and the choice lies between acetylene from the EPIDC natural gas, based plant, or, the use of some of the ethylene from Fauji's naphtha cracker. In this situation, the acetylene route to vinyl acetate is clearly the cheaper.



: L. VOLUME 11 4-4

| HUMPHREYS & GLASGOW | Volume II 4 - 5 | |
|---------------------|---|-----------|
| UNIDO Vienna | Pre-Investment Studies for Pertiliser | C.1669 |
| for Pakistan å | Petrochemical Industries - Final Report | July 1970 |

4.2 RECONDENDED DEVELOPMENT AND FINANCIAL EVALUATION FOR ASHUGANJ SITE, EAST PAKISTAN

4.2.1 Introduction

The recommended development for Phase II of the Ashuganj project is shown in the Block Diagram II 4.1. Section 4.1 discusses the reasons underlying its composition.

The layout of this section follows that of Section 3.2, dealing first with the individual process units, then the development scheme as a Subsequent sections deal with capital and operating costs. whole. Finally, the section concludes with the Foreign Exchange Analysis and the year-by-year Cash Flows, including the D.C.F. rate of return on the investment.

4.2.2 Individual Process Units

4.2.2.1 Acetaldehyde

Process Description and Flow Chart

A simplified flow chart for the production of Acetaldehyde is shown in Figure II 4.2.

Acetaldehyde is produced by direct hydration of acetylene, catalysed by mercuric ions. Acetylene is injected into mercuric sulphate solution, acidified by sulphuric acid, contained in the reactor. Acetaldehyde vapours mixed with steam, acetic acid and unreacted acetylene from the reactor are cooled and passed to an absorber where the acetic acid is absorbed by the aqueous acetaldehyde solution from the bottom of the next absorber. The gas from the acetic acid absorber is further cooled before passing to the acetaldehyde absorber mentioned above, and the unreacted acetylene, together with some inerts, is recycled back to the reactor, while a small portion is continuously purged to maintain the inerts level. This purge, still containing some acetylene, is returned to acetylene plant for acetylene recovery.

The aqueous acetaldehyde solution from the absorption system flows through a mercury settling tank, and after filtration is preheated before entering the first distillation tower from which water and byproducts are removed as a bottom stream. This stream, after heat recovery, is sent to drain. The distilled acetaldehyde is finally purified in the acetylene stripper, leaving as a bottom stream of 99.8% purity. The uncondensed overhead gases from both towers contain acetaldehyde and are recycled to the bottom of the mostaldehyde absorber.



-

ł

| | BOW LTD. | Volume II 4 - 7 | |
|------------------------------|--|---------------------|--|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fsrtiliser & Petrochemical Industries - Final Report | C.1669 July 1970 | |

The mercuric sulphate catalyst solution is continuously regenerated by hot treatment with nitric acid in the presence of an air stream blown into the solution. Mercury-containing muds from this catalyst regeneration plant are sent to mercury retorts for the recovery of metallic mercury. This part of the process is not included in the flow chart.

Source of Know-how and Licence

The process as described above is available from Montecatini Edison Spa. A similar process is available from Lonza Ltd.

Operating Data

Table II 4.2 below lists the major consumptions of raw materials, catalysts/chemicals, utilities and direct labour requirements.

| OFERATING DATA ~ ACETALDEHYDE | (BASIS 1 M ACETALDEHYDE) |
|-------------------------------|--------------------------|
| Acstylene (as 100%) MT | 0.63 |
| Catalyst/Chemicals \$ | 4.8 |
| Power, KWH | 300 |
| Steam (low pressure) MT | 2.2 |
| Cooling Water M ³ | 9 0 |
| Labour Requirements | 18 |

Area of Battery Limits

The approximate area of the battery limits plant is 1250 m^2 . This is based on a layout of the plant items, including those omitted from the flow chart.

4.2.2.2 2-Ethyl Hexanol

Process Description and Flow Chart

A simplified flow chart for the production of 2-ethyl hexanol is shown in figure II 4.3.

The first part of the process may be considered as the preparation of crotonaldehyde. Acetaldehyde from the acetaldehyde plant, together with recycle from the process, is mixed with dilute caustic soda catalyst in the reactor mixer. The condensation reaction is arrested by the introduction of acetic acid near the base of the column as the reaction mixture flows downwards and thence to the reactor

SECTION 1

PIG 11.4 3









. .

| HUMPHREYS & GLASGOW LTD. | | Volume II 4 - 9 |
|--------------------------|---|--------------------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

column. Here the aldol is converted into crotonaldehyde under the influence of pressure and temperature. The overhead vapours from this column are cooled and pass to a distillation column. Here acetaldehyde is recovered overhead and recycled, while from the base a mixture of water and crotonaldehyde is withdrawn and separated by cooling and decantation. Crotonaldehyde is recovered from the water rich phase in a small stripping column at the base of the crotonaldehyde column.

The second part of the process involves the preparation of butyraldehyde. Crotonaldehyde from the previous part of the process is first used to wash the hydrogen purge stream. Tt then passes to the reactor into which is blown a large excess of hydrogen and which is heated by steam coils in the base. Overhead reaction products pass to the butyraldehyde column Butyraldehyde is withdrawn from (which has no reboiler). this column as a side stream, while heavy ends are withdrawn from the base. The overhead condensate is separated into two phases, the aqueous phase being next to a water stripper. The overhead hydrogen is recycled after further cooling. The hydrogen purge, required to maintain the inerts level, is washed with the feed crotonaldehyde and then with water which is stripped of organics in the water stripper. The recovered organics pass to the reactor.

The third part of the process is the preparation of 2-Ethyl Hexanol. Butyraldehyde from the previous stage is introduced into a circulating reaction system with an alkaline catalyst. The conversion to the hexanal is complete, and after decantation from the water formed, it is vapourised, mixed with hydrogen and passes through catalyst packed tubes of the hydrogenation reactor. The heat of reaction is used to generate steam in the shell side of the reactor. The reaction mixture is cooled, and the separated hydrogen reheated and recycled to the hydrogenation stage. The crude Ethyl Hexanol passes to two distillation columns, which remove the secondary products, and then to a finishing hydrogenation The stream passes through two further distillation reactor. columns which remove the last traces of secondary products. The Ethyl Hexanol is suitable for the production of plasticisers. The concurrent production of butanol cannot be avoided and is at least 70 Kg per ton of Ethyl Hexanol.

Source of Know-how and Licence

The process as described above is available from Melle-Bezons S.A.

HUMPHREYS & GLASGOW LTD.



| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
|--------------|---|-----------|
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

Operating Data

This is given below in Table II 4.3.

<u>TABLE II 4.3</u> OPERATING DATA - 2 ETHYL HEXANOL (BASIS IMT 2 ETHYL HEXANOL)

| Acetaldehyde (100%) MT | 1.66 |
|------------------------------|---------------|
| Hydrogen (100%) NM3/Kg | 830/74.5 |
| Catalysts/Chemicals \$ | 16.8 |
| Power, KWH | 406 |
| Steam (high pressure) MT | 0.14 |
| Steam (medium pressure) MT | 11.15 |
| Steam (low pressure) MT | 2.96 (CREDIT) |
| Cooling Water M ² | 300 |
| Labour Requirements | 30 |

Area of Battery Limits

The approximate area of the battery limits plant is 1450 m². This is based on a layout based on the flow chart, but including for those plant items not included in the chart.

4.2.2.3 Dioctyl Phthelate

Process Description and Flow Chart

A simplified flow chart for the production of Dioctyl Phthalate is shown in figure II 4.4.

Phthalic anhydride is dissolved in 2 ethyl hexanol from the ethyl hexanol plant and flows, together with sulphuric acid catalyst, to the top of the esterifier. Cyclohexame is used as a water entrainer and is vapourised and introduced into the bottom of the esterifier.

The vapour from the top of the esterifier, together with that from the dissolving vessel, pass to a separator. The overheads from this separator are condensed and separated into two phases. The alcohol rich phase is returned as reflux to the separator, and after contacting the upflowing vapours is recycled to the dissolving tank. The cyclohexane and water phase still containing some alcohol, pass to the alcohol and cyclohexane recovery columns.

The crude ester leaves the bottom of the esterifier and enters a steam heated prestripping column, from the top of which the bulk of the cyclohexane is removed. The cyclohexane



FIG. 1.4.4

U.N.I.D.O. - C-1669 FLOW CHART - DIOCTYL PHTHALAT E.P.I.D.C. - ASHUGANJ - EAST PAKISTAN -



HUMPHREYS & GLASGOW LTD.

Volume II 4 - 12

| UNIDO Vienna | Pre-Investment Studies for Fertilizer | C.1669 | |
|--------------|---|-----------|--|
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

phase which separates after condensation is passed to the alcohol/cyclohexane recovery system. The ester, after further heating, next passes to a neutralising column where the added sodium carbonate completely neutralises the acid catalyst and ensures the thermal stability of the dioctyl phthalate. The stream is then cooled, and the water which separates flows to drain. The ester stream next passes to column in which water, separated in a downstream column, washes out the neutralising sodium carbonate and then also flows to drain. The alcohol is then steam stripped from the ester, and is concentrated in the alcohol rich phase obtained after condensation of the overheads and decantation. This also passes to the alcohol/cyclohexane recovery system. The ester and water, leaving the bottom of the stripper are separated into two phases, the water rich phase being used to wash the crude ester in the preceding wash column, and the ester being passed to a column where it is finally dried with air before passing to storage. The air from the top of this column is cooled before venting to recover the ester.

The alcohol recovery column recycles ethyl alcohol from the bottom while light secondary products are taken overhead. A sidestream from this column takes the cyclohexane to the cyclohexane column: heavy ends are taken from the bottom while water is separated overhead. Cyclohexane is removed as a sidestream and recycled to the start of the process. The feature of sidestream removal arises because of the presence of azeotropes and the fact that the maximum cyclohexane concentration occurs in the liquid phase from the upper trays.

Source of Know-how and Licence

As for 2-ethyl hexanol, the process technology is available from Melle-Besons S.A. Comparable technology is available from B.P. Chemicals Ltd.

Operating Data

The essential data is given below.

| HUMPHREYS & GLA | SGOW LTD. | Volume II 4 - 13 | _ |
|-----------------|---|---------------------|---|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

TABLE II 4.4 OPERATING DATA - DIOCTYL PHTHALATE (BASIS 1MT DIOCTYL PHTHALATE)

| 2 Ethyl Hexanol (as 100%) MT | 0.73 |
|---------------------------------|-------|
| Phthalic Anhydride (as 100%) MT | 0.385 |
| Catalysts/Chemicals \$ | 2.4 |
| Power, KWH | 25 |
| Steam (medium pressure) MT | 1.9 |
| Cooling Water, M ² | 66 |
| Process Water, M ³ | 0.4 |
| Labour Requirements | 18 |

Area of Battery Limits

The approximate area of the battery limits plant is 760 m^3 . This was obtained by a layout based on the flow chart, but including those items omitted from the chart.

4.2.2.4 Acetic Acid

Process Description and Flow Chart

A simplified flow chart for the production of the Acetic Acid is shown in figure II 4.5.

Acetic acid is produced by air oxidation of acetaldehyde, in the presence of manganese acetate catalyst. Acetaldehyde from the acetaldehyde plant, and recycled acetaldehyde, is charged to the oxidation reactor into which passes air from the Air Separation Plant after water washing in the direct contact coolers. The reaction temperature is maintained by circulating the mixture through an external cooler.

The crude acetic acid from bottom of the reactor passes to the peroxide breaker, where the explosive peracetic acid concentration is reduced by long residence at high temperatures. The reactor off-gas is scrubbed first with crude acetic acid to recover acetaldehyde and then with water to recover acetic acid. The crude acetic acid wash stream is returned to the reactor, and the water wash stream passes to the methyl acetate column.

From the peroxide breaker, crude acetic acid is fed to the scetaldehyde column. Unreacted acetaldehyde is recovered overhead and recycled to the acetaldehyde surge drum. The bottom stream from this column is cooled, part being used to wash the off-gas as described above, and the remainder is charged to the methyl acetate column, from which a net



HUMPHREYS & GLASGOW LTD.

UNIDO Vienna for Pakistan

Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report

4 - 15

Volume II

C.1669 July 1970

methyl acetate stream is taken overhead. The bottom stream from the column is charged to the acetic acid finishing column, from the bottom of which glacial acetic acid of 99.8% purity is withdrawn (actually withdrawn from above bottom plate as a vapour and the bottom liquid stream, containing manganese acetate catalyst is recycled to the reactor. This has not been shown in the flow chart).

The overheads from the finishing column is an azeatropic mixture of ethyl acetate and water. Condensation and decantation separate two layers: the ethylacetate rich phase is used as reflux, while the aqueous phase is fed to the ethyl acetate column. The overhead azeotrope from this column is returned to the condensation/decantation stage described above, while the bottom stream from this tower is waste water, and is sent to drain.

Source of Know-how and Licence

This process is a well known chemical process which has been operated by many major companies. Melle-Bezons S.A., British Petroleum and Hoescht are suggested as potential licensors. <u>Operating Data</u>

_

The essential data is given below.

TABLE II 4.5 OPERATING DATA - ACETIC ACID (BASIS INT ACETIC ACID)

| Acetaldehyde (as 100%) MT | 0.77 |
|------------------------------|------|
| Power, KWH | 260 |
| Steam (low pressure) MT | 3.58 |
| Cooling Water M2 | 300 |
| Process Water M ³ | 13 |
| Nitrogen NM3 | 15 |
| Labour Requirements | 18 |

(Note: Catalysts and chemicals used in the process are about equal in valve to the methyl acetate by-product.)

Area of Battery Limits

The approximate area of the battery limits plant is 480 m². This was obtained by a layout based on the flow chart, including those items omitted from the chart.

HUMPHIEVS & GLASGOW LTD.

Volume II 4 • 16

UNIDO Vienna for Pakistan &

Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report

C.1669 July 1970

4.2.2.5 Vinyl Acetate

Process Description and Flow Chart

A simplified flow chart is given in figure II 4.6.

Acetic acid from the acetic acid plant is vapourised and mixed with acetylene, some of which is recycled from the process. Acetic acid vapour recycled from the process joins the feed stream after the evaporator. After preheating the acetic acid and acetylene vapours pass to a fluidised-bed type reactor, which is cooled both internally and externally by (for example) Mobiltherm Light. A portion of catalyst is taken continuously from the bottom of the reactor to a regenerator (via a storage vessel which is not shown). Here it is subjected to blowing with superheated steam, hot inert gas and zinc acetate and active carbon are replenished. The regenerated catalyst is returned continuously to the reactor via another storage vessel (not shown in flow chart). Vent gas from the regenerator is cooled before joining the recycle acetylene stream. The reaction products pass through a bag filter to remove catalyst dust, are cooled and then contacted with cold crude vinyl acetate in two cyclonic spray towers in In these vinyl acetate and acetic acid are condensed series. to form crude vinyl acetate. Non-condensible gases, (mainly acetylene) are recycled, with a small portion being vented after scrubbing with crude acetic acid to control the level of inert gases. The scrubbing stream passes to the condensing system.

Crude vinyl acetate is degassed, and the acetaldeyde condensed from the overheads. This contains some vinyl acetate which is separated in the vinyl acetate column and recycled to the degasser column. Acetaldehyde is returned to the acetaldehyde plant. Non-condensibles from the two columns, consisting mainly of acetylene, are recycled.

The vinyl acetate stream from the bottom of the degasser passes to a column from pure vinyl acetate is taken overhead, and acetic acid from the bottom. Two side-streams are drawn from this tower. The one from the top section contains mainly divinylacetylene, and the one from the bottom section contains crotonaldehyde. These are distilled separately in two smaller columns. They may be treated separately in a batch still to further recover vinyl acetate and acetic acid.

Crude acetic acid from the vinyl acetate column contains appreciable amounts of heavy impurities. These are removed by passing part of the crude acetic acid to the acetic acid column. The pure acetic acid is recycled to the beginning of the process. Polymerisation inhibitor is added to the








]

CHUICANNE CHUI FARIUNA FAN

| HUMPHREYS & GL | NOW LTD. | | Volume II 4 _ 1 |
|------------------------------|--|--|--|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for & Petrochemical Industries -) | Fertiliser Final Report | C.1 669 Jul y 1970 |
| | | | |
| | degassing and vinyl acetate co | olumna | |
| | Source of Know-how and Licence | 2 | |
| | This process is well establish to be readily available. Bri Hoechst are proposed as possib | hed technology as itish Petroleum, ple licensors. | nd is expected I.C.I. and |
| | Operating Data | | |
| | The essential data is given be | low- | |
| | TABLE II 4.6 | | |
| | OPERATING DATA-VINYL ACETATE (B | MSIS 1 MT VINYL | ACETATE) |
| | Acetylene (as 100%) MT Acetic Acid (as 100%) MT Catalyst/Chemicals \$ | 0. 32 0.72 1.8 | |
| | Power kWh Steam (medium pressure) MT Cooling Water M ² Inert Gas NM ³ | 110 2.28 120 150 | |
| | Labour Requirements | 18 | |
| | Area of Battery Limits | | |
| | The approximate area of the bat 640 M ² . This was obtained by items. | ttery limits play a layout of the | nt is plant |
| 4.2.2 | .6 Acetylene | | |
| | The extension of the acetylene to 55,150 MTPA (in 300 days per the extens on of the ethylene p extremely likely that the make- of the plant will be designed f required by Phase II, and that will be added to the existing p time. | plant from 36,00 year), is anala plant at Fauji. gas treatment se for the larger ca add tional burne plant at the appr | O MTPA gous to It is ection pacity r units opriate |
| | In view of this fact, the proce | ss description, | flow |

|) | | NOW LTD. | Volume II 40-19 |
|---|--------------|---|---------------------------|
| | UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

data is required, however, for evaluating the variable and average costs of production for use in later sections. This data is given below :-

TABLE II 4.7

OPERATINO DATA-ACETYLENE (BASIS 1 MT ACETYLENE)

| Natural Gas, scf. | 214,000 |
|---|---------|
| Catalysts/Chemicals \$ | 1.0 |
| Byproduct Off-gas (credit) K cal x 10 ⁰ | 27.8 |
| Power kWh | 3290 |
| Steam (net) MT | 3.9 |
| Cooling Water M ² | 870 |
| Process Water (net) M ³ | 1.3 |
| Increase in | |
| labour requirements (36,000 to 55,000 MTPA) | 5 |
| A Description of A Delivery | |

4 2.3 Complete Development Scheme

4.2.3.1 Utilities and Offsites Installation

With regard to these installations for the E.P.I.D.C. scheme at Ashuganj, the same arguments already discussed in Section 3.2.3.1 of this report apply.

It was therefore considered that the most realistic costs of the various utilities required for Phase II would be those based on a utilities installation of a similar size to that required for the complex after the addition of Phase II. Based on previous work for the Interim Report, such utilities costs have been evaluated for the Ashuganj site; these are derived from unit capital costs and operating data in a similar way to the derivation of petrochemica' production costs. They are based on a fully integrated utilities scheme, and are given in Section 4.2.5

| HUNTHREYS & QLA | Volume II 11 - 20 | | |
|-----------------|---|----------------|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C .1669 | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

4.2.3.2 Utilities Consumption

The following table summarises the main utilities consumption required by the recommended development for Phase II :-

TABLE II 4.8

SUMMARY OF TOTAL UTILITIES CONSUMPTION FOR PHASE II

Power kw12,000Steam MT/hr47.2Cooling Water M³/hr4,140Condensate/Demin. Water (for process)
M³/hr21.1Inert Gas (Nitrogen from A.S.U.)
NM³/hr212

4.2.3.3 Management, Personnel and Labour

The direct process labour requirements for Acetaldehyde, Ethyl Hexanol, Dioctyl Phthalate, Acetic Acid and Vinyl Acetate Plants are 107 men. A further 5 men are allowed for the expansion of the Acetylene Plant, and this number is considered adequate to cover any additional requirements of a possible extension to the Utilities Installation.

Plant Managers and partial supervision at higher levels are not included. Their cost is covered by the use of a weighted rate of 9,000 Rs/year for each man, as explained in Section 3.2.3.3 of this report for the Fauji Foundation. Each of new plants above would require a Plant Manager, i.e., 5 in all.

4.2.3.4 Plot Plan of the Recommended Development

Figure II 4.7 gives the plot areas of the process plants a possible arrangement. A tank farm of approximately 2,200 M² area for the storage of liquid products is included in the plan.

The disposal of effluents has been viewed in a similar way to the Fauji Foundation scheme (see Section 3.2.3.4) and it is assumed that either the Phase I Effluent Disposal System will be capable of handling the additional load, or will be extended as necessary. HUNPHREYS & GLASGOW LINITED

4

VOLUME II 4-2.1



YOUTHE

.

| NUMPHREYS & (| LASCOW LTD. | | | | Volume II 4 - 22 |
|---|---|--|---|--|--|
| UNIDO Vienna for P akistan | P re- & Petr | Tnvestment Stu Pochemical Indu | dies fo stries | r Fertiliser - Einal Report | С .1669 July 1970 |
| 4.2.3.5 | Use of Acetyle: | ne Off-gas | | | |
| | It is j acetylene plant This is confirm being self-supp normal operation It can consumption in capacities spec | bossible to make t as fuel within and by the foll porting in stea on. be assumed for normal operation cified in the A | te use o in the F lowing e im and e Phase lon will IPC Stud | of all the off-gas EPIDC petrochemica estimates, based of electrical power g I that steam and I be about 85% of By document (refer | e from the al complex. on the complex generation in electricity the design mence 45A), |
| | nence:- Phase I | HP Steam | 51 M | T/hr | |
| | | LP Steam | 161 M | T/hr | |
| | | Electricity | 51,0 | 000 kw | |
| | By addi as already atat consumption fig | ing to these fi ted in Table II gures for the c | gures t 4.2 ab | the Phase II consu pove, we obtain, th d complex: - | mption rates e following |
| | Phase I & Phase | II HP Stea | a | 51 MT/hr | |
| | | LP Stea | A | 208 MT/hr | |
| | | El cobred | | 67 000 1 | |

The total process steam requirement of 259 MT/hr, all raised igitially at high preasure, will require approximately 200 x 10 kcal/hr of fuel energy, while approximately 27,000 kw would be obtained by passing-out the LP Steam requirement of 208 MT/hr from turbo-alternator sets. A further quantity of steam would be raised, and eventually condensed at subatmospheric pressure, to generate the balance of electrical power required, i.g. 36,000 kw. The additional steam would require about 100 x 10 kcal/hr of fuel energy. Hence, the total fuel required for steam and electricity generation is equivalent to about 300 x 10 kcal/hr. Additional fuel will be required to fire the steam reforming furnaces making synthesis gases for the ammonia and methanol plants, only a small part of which is satisfied by purge gases from the synthesis plants.

The off-gas rate (based on 55,000 MTPA of acetylene in 300 days/year operation and the operating data of Table II 4.7 above) is equivalent to $217 \times 10^{\circ}$ kcal/hr. Hence all the off-gas can be used as fuel.

APC proposed that the off-gas be used as feedstock for the ammonia/urea and methanol process plants. We estimate that there would be sufficient off-gas available to satisfy the process feedstock requirement of the methanol plant, and a large part of the feedstock requirement for the ammonia/urea plant as well.

The advantage gained by such use of the off-gas is the partial or complete elimination of the natural gas reforming step in the

| | Volume II 4 - 23 | |
|--------------|---|-------------------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 19 70 |

o

ammonia and methanol synthesis processes. There are a number of possible ways in which the normal synthesis processes could be modified and integrated with the acetylene/oxygen plants so that such a utilisation of the off-gas is achieved. There are, however, two major - and in our view over-riding - disadvantages in such an integration. The first is the fact that, while natural gas is available at a pressure in excess of 1200 psi the acetylene plant off-gas is only marginally above atmospheric pressure. Thus the cost of gas compression in capital and operating costs is higher if acetylene off-gas is used for the synthesis processes. Secondly, it seems very unwise to link such large complexes in this way. The ammonia and methanol plants would both suffer the downtime of the acetylene complex as well as downtime resulting from their own internal failures. This could easily mean 10% additional downtime equivalent to the design requirement of 10% higher daily capacity for the complete urea and methanol sections of the plant. Similarly off-gas from the acetylene plant would be wasted when the ammonia or methanol plants were out of operation. We consider, however, that it would be possible to design the two synthesis plants to use the acetylene off-gas as fuel if this is shown to be attractive.

| | | Volume II | |
|---------------------|---|-----------|--|
| HUMPHREYS & GLASGOW | LTD. | 4 - 24 | |
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | |
| for Pakistan & | Petrochemical Industries - Final Report | July 1970 | |

4.2.4 Capital Costs

4.2.4.1 Acetaldehyde Plant

TABLE II 4.9

CAPITAL COST OF ACETALDEHYDE PLANT

| | | Rupees x | Rupees x 10 ⁶ | |
|----|--|---------------------|--------------------------|--|
| 1. | Equipment, Spares, Procurement. | Foreign Exchange | Local Currency | |
| | Freight and Insurance, Tools and Plant, Insurance. | 12.94 | - | |
| 2. | Engineering, Expenses and Overheads, Licence, Site Super- vision and Commissioning. | 6.20 | - | |
| 3. | Construction Labour, Civil work, Local tools and plant, Local insurance, Inland Freight and Dock clearance, Local Spares, | | | |
| | Local Expenses. | - | 4.98 | |
| 4. | Duty on Equipment and Spares | • | 3.13 | |
| | TOTALS: | 19.14 | 8.11 | |
| | Start-up charges Interest on Foreign Exchange | 1.36 | 1.36 | |
| | during construction | 2.11 | • | |
| | | 22.61 | 9.47 | |
| | TOTAL COST | 32.0 | 3 | |

The above capital cost estimate covers all fixtures which are normally within the plant battery limit, and all expenses required to bring the plant into operation.

<u>Included</u>:- Raw materials and products storage and handling sufficienct for normal operation. Civil work, control rooms, minor office buildings and workshops, fire equipment, telephones etc. within the plant boundary, as necessary.

Excluded: - Major offsite storage facilities, Civil work, piping, and services distribution up to the plant boundary, site development, and land.

Interest during construction on the local currency element is omitted from this estimate because details of the method of finance are not known.

| HUMPHREYS & GLASGOW LT | 2 | Volume II 4-25 |
|------------------------|---|-------------------|
| UNIDO V ienna | Pre-Investment Studies for Fertiliser | 0.1669 |
| for Pak istan & | Petrochemical Industries - Final Report | July 1970 |

4.2.4.2 2-Ethyl Hexanol Plant

TABLE II 4,10

CAPITAL COST OF 2-ETHYL HEXANOL PLANT

| | | Rupees x | <u>10</u> 6 |
|----|---|---------------------|-------------------|
| | | Foreign Exchange | Local Currency |
| 1. | Equipment, Spares, Procurement, Freight and Insurance, Tools and Plant, Insurance. | 6.51 | - |
| 2. | Engineering, Expenses and Overheads, Licence, Site Supervision and Com- missioning. | 4.64 | - |
| 3. | Construction Labour, Civil work, Local tools and plant, Local insurance, Inland Frieght and Dock clearance, Local Spares, Local Expenses. | | 2.33 |
| 4. | Duty on Equipment and Spares | - | 1.54 |
| | TOTALS : | 11.15 | 3.87 |
| | Start-up charges Interest on Foreign Exchange | 0.75 | 0.75 |
| | during construction | 1.23 | • |
| | | 13.13 | 4.62 |
| | | | |

For definition of the scope of the above capital cost, see footnote to Table II 4.9.

| | IOW LTD. | Volume II 4-26 | |
|------------------------------|--|-----------------------------|--|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C. 1669 July 1970 | |

4.2.4.3 Dioctyl Phthalate Plant

TABLE II 4.11

CAPITAL COST OF DIOCTYL PHTHALATE PLANT

| | | Rupees x 10 ⁶ | | |
|----|---|--------------------------|-------------------|--|
| | | Poreign Exchange | Local Currency | |
| 1. | Equipment, Spares, Procurement, Prieght and Insurance, Tools and Plant, Insurance. | 6.08 | - | |
| 2. | Engineering, Expenses and Overheads, Licence, Site Super- vision and Commissioning. | 2,50 | - | |
| 3. | Construction Labour, Civil work, Local Tools and Plant, Local insurance, Inland Freight and Dock Clearance, Local Spares, Local Expenses. | - | 2.31 | |
| 4. | Duty on Equipment and Spares | - | 1.47 | |
| | TOTALS : | 8.58 | 3.78 | |
| | Start-up charges Interest on Foreign Exchange | 0.62 | 0.62 | |
| | during construction | 0.94 | - | |
| | | 10.14 | 4 : 40 | |
| | TOTAL COST ESTIMATED WORKING CAPITAL (max.) | 1 | 4.54 7.47 | |

For definition of the scope of the above capital cost, see footnote to Table II 4.9.

| | LTD. | | volume I 4-2 |
|------------------------------|--|---------------------|---------------------------|
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Ferti & Petrochemical Industries - Final | lliser L Report | C.166 July 197 |
| 4.2.4.4 | eetic Acid Plant | | |
| - | TABLE II 4.12 | | |
| | CAPITAL COST OF ACETIC ACID PLAN | 2 | _ |
| | | Rupees : | <u>× 10⁶ (</u> |
| | | Foreign Exchange | Local Currency |
| : | . Equipment, Spares, Procurement, Freight and Insurance, Tools and Plant, Insurance. | 5.51 | • |
| 4 | 2. Engineering, Expenses and Over- heads, Licence, Site Supervision and Commissioning. | 1.53 | - |
| : | 5. Construction Labour, Civil wrok, Local Tools and Plant, Local insurance, Inland Freight and Deck Clearance, Local Spanse | | |
| | Local Expenses. | • | 2,08 |
| | . Duty on Equipment and Spares | - | 1.33 |
| | TOTALS : Start-up charges | 7.04 | 3.41 0.52 |
| | Interest on Foreign Exchange during construction | 0.77 | |
| | | 8.33 | 3.93 |
| | TOTAL COST ESTIMATED WORKING CAPITAL (max.) | 12 0. | . 26 . 35 |

J

1

١.

| 4-28 | W LTD. | UNPHNEYS & GLASGI |
|-----------|--|-----------------------|
| C.1669 | Pre-Investment Studies for Pertiliger | UNIDO Vienna |
| July 1970 | & Petrochemical Industries - Final Report | for Pakistan |
| | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | DO Vienna Pakistan |

CAPITAL COST OF VINYL ACETATE PLANT

Rupees x 10⁶

| | | Foreign Exchange | Looal Currency |
|----|--|----------------------|-------------------|
| 1. | Equipment, Spares, Procurement, Freight and Insurance, Tools and Plant, Insurance. | 1 6.18 | - |
| 2. | Engineering, Expenses and Over- heads, Licence, Site Supervision and Commissioning. | n 1.83 | - |
| 3. | Construction Labour, Civil work, Local Tools and Plant, Local insurance, Inland Freight and Dock Clearance, Local Spares Local Expenses. | - | 2.35 |
| 4. | Duty on Equipment and Spares | - | 1.50 |
| | _ | | متزاكرته |
| | TOTALS : Start-up charges Interest on Foreign Exchange | 8.01 0. 59 | 3.85 0.59 |
| | during construction | 0.88 | - |
| | | 9.48 | 1.14 |
| | TOTAL COST ESTIMATED WORKING CAPITAL | | 13.92 2.49 |

For definition of the scope of the above capital cost, see footnote to Table II 4.9

| f U | NIDO Vienna pr Pakistan | Pre-Investment Studies for Fer & Petrochemical Industries - Fir | tiliser Mal Report | C.1669 July 1970 |
|-----|----------------------------|--|---|--|
| ſ | 4.2.4.6 | Expansion to Acetylene and Oxygen F Utilities and Offsites | lants, | |
| | | The cost of expanding the productic Acetylene and Oxygen plant from 36, been estimated, and is given, in Ta Allowances for expansion of Utility given. | n capacity c OOO MTPA to able II 4.14 es and Offsi | f the 55,150 MTPA has below. tes are also |
| | | TABLE II 4.14 | | |
| | | CAPITAL COSTS OF ACETYLENE. | UTILITIES | |
| 1 | | AND OFFSITES EXPANSION | NS | |
| • | | | Rupees x | <u>10</u> 6 |
| | · · · | | Foreign Exchange | Local Currency (Inc. Duty) |
| | | Acetylene and Oxygen Plants Expansion | 14. 3 9 | 8.71 |
| 1 | | Utilities Plants Expansion | •75 | .46 |
| | | Offsites Expansion | 3.45 | 6.91 |
| I | | Interest on Foreign Exchange During Construction | 1.98 | - |
| | | TOTALS : | 20.57 | 16.08 |
| - | | | _ | |

۱. .

| | IGOW LTD. | Volume II 4-30 |
|--------------|---|-------------------|
| UNIDO Vienna | Pre-Investment Studies for Pertiliser | C.1669 |
| IOF PARISTAN | & Fetrochemical Industries - Final Report | July 197 0 |

4.2.4.7 Capital Costs Summary for Expension

| TABLE II.4.15 Capital Costs Summary for Expansion of Acetylene Complex | | | | |
|---|------------------|--|---------------------------------|--|
| | Capacity MyPA | Rupees x 10 ⁶ Foreign Exchange | Local Currency (Inc.Duty) | |
| Acetaldehyde Plant | 25,90 0 | 19.14 | 8.11 | |
| 2-Ethyl Hexanol Plant | 11 ,84 0 | 11.15 | 3.87 | |
| Dioctyl Phthalate Plant | 16,00 0 | 8.58 | 3.78 | |
| Acetic Acid Plant | 8,18 0 | 7.04 | 3.41 | |
| Vinyl Acetate Plant | 8,420 | 8.01 | 3.85 | |
| TOTALSI | | 53.92 | 23.02 | |
| Start-up, training and consultancy costs | | 3.84 | 3.64 | |
| Interest during construct | tion | 6.22 | - (2.53) | |
| TOTALS FOR NEW PLANTS: | | 63.98 | 26.86(29 .39) | |
| TOTALS FOR ACETYLENE, UT AND OFFSITES EXPANSION | ilitiks Is | 20.57 | 16.08 | |
| TOTALS FOR EXPANSION OF | COMPLEX | 84.55 | 42.94 | |
| TOTAL COST | | 127.49 | | |
| ESTIMATED WORKING CAPITA | L | 10.31 | | |

The 'totals for new plants' are needed for the cash flow calculation in Section 4.2.7.6 in which the average unit cost of acetylene includes an element to cover the capital costs of expansion of the acetylene and oxygen plants, utilities and offsites.

For the cash flow calculation, the figures used for local currency costs are shown in brackets. 'Interest' amounting to 11% of the local currency element of plant cost is included, to allow for the time value of money during the construction period.

| HUMPHREYS & GLASG | OW LTD. | Volume II 4-31 | |
|-------------------|---|-------------------|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

4.2.4.8 Pre-Investment for Phase II

It will be necessary to lay out a significant proportion of the capital cost of the expansion in the initial construction stage, so as to achieve the most economical result overall.

The acetylene purfication equipment would be designed and erected as a single stream for the final (55,150 MTPA) capacity. Six burner units would be required for this capacity and of these, two would be omitted from their allotted places until the second stage of construction, together with the associated gas scrubbing and cooling equipment.

The cxygen plant and the main utilities plants would be installed in their final form in the first stage of construction, apart from a few items, such as a cooling water cell, which could be delayed.

The 'offsites' (utilities distribution etc.), attributable to the expansion, would also nearly all be installed during the first construction phase.

4.2.4.9 Capital Cost Summary for Project

In the following table, the estimate of the capital cost for Phase I, as sanctioned, is taken from EPIDC's proposal document dated January 1970. The additional capital cost for the recommended expansion has been calculated by adding together the estimates already given for the process plants and for the expansion of the acetylene and oxygen plants, utilities and offsites. Of this additional capital, the pre-investment as described above has been estimated, and is given below under the heading Phase II A. The remainder of the capital cost, necessary to complete the recommended expansion, is given under the heading Phase II B.

| | IOW LTD. | Volume II 4-32 | |
|--------------|---|-------------------|--|
| UNIDO Vienna | Pre-Investment Studies for Pertiliger | C.1669 | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

TABLE 11.4.16

Capital Cost Summery for EPIDC Petrochemical Project

| | Re. x | <u>100</u> |
|---|-------------------|------------------------------|
| | Foreign Exchange | Local Currency (Inc.Duty) |
| Phase I as sanctioned (including land, site development and | | |
| housing colony): | 562 | 395 |
| TOTAL COST: ESTIMATED WORKING CAPITAL: | 957 12 | 7 |
| Phase IIA as defined above: | 18.51 | 15.81 |
| TOTAL COST: | 3 | |
| Hence, Stage 1 of construction: | 58 0.51 | 410.81 |
| TOTAL COST: ESTIMATED WORKING CAPITAL: | נ פפ גל | |
| Phase IIA (Stage 2 of construction | an): 66.04 | 27.13 |
| TOTAL COST: ESTIMATED WORKING CAPITAL | 93 10 |).17).31 |
| Complete Project: | 646.55 | 437.94 |
| TOTAL COST: ESTIMATED WORKING CAPITAL | 1,084 | .49 |

| | | Volume II |
|------------------------------|--|---------------------|
| HUMPHREYS & GLAS | GOW LTD. | 4-33 |
| UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 July 1970 |

4.2.5 Unit Costs and Prices

TABLE II 4.16 lists the unit costs of the various raw materials, utilities and packages used in the subsequent financial analysis. It also gives the credit values assigned to the various by-products, and the proposed selling price of the end-products. These costs, credit values and prices are sub-divided for the purpose of specific financial analyses.

÷¥

| TABLE | II | 4. | 17 |
|-------|----|----|----|
| | _ | | |

| UNIT COSTS A | ND PRICES | | D - |
|---|--------------------------|--|--------------|
| End Products (\$) | ¥ Foreign Exchange | rs. Local Currency Excl. Dutr | RE. Total |
| Dioctyl Phthalate, MT (235) | 38 4 | النظيفة المتعقبة - | 2,800 |
| Acetic Acid, MT (107) | 161 | • | 1,280 |
| Vinyl Acetate, MT (149) | 222 | • | 1,775 |
| By-products | | | |
| Off-gas. 10 ⁶ K.Cal. | • | 2.34 | 3.89 |
| | | - | |
| TAN MALUFIALE | | | |
| Natural Gas, 10' s.c.f. | • | 0.6 | 1.0 |
| Phthalic Anhydride, MT 1. indi | encus- | 1510 | 1510 |
| 2, 1 mp 01 | rted 270 | 100 | 100 |
| hydrogen, MT | • | 20 | 20 |
| <u>Utilities</u> | | | |
| Electricity, kWh | 0.005 | 0.035 | 0.063 |
| Steam, MT | 0.176 | 2.23 | 4.90 |
| Cooling Water, m2 | 0.004 | 0.048 | 0.075 |
| Process Water, m2 | 0.071 | 0.43 | 1.00 |
| Demin. Water, m ² | 0.22 | 0.85 | 2.68 |
| Packages | | | |
| Steel Drums (per MT of D.O.P. or V.A.) | • | 60 | 60 |
| Polythene Containers (per MT | • | 30 | 30 |

Note on Prices

The "total" prices for the end products are the proposed selling prices before tax estimated by H & G on the basis that they would give a reasonable return on the capital employed, suitably distributed over these end products. The cash flow calculation provides an overall check on these estimates. The foreign exchange prices of these end products are the expected C & F prices of the substituted imports, and they are based upon current C & F prices. The foreign exchange prices shown in brackets are an estimate of the C & F price which would be competitive, under cash cum bonus, with the proposed selling price.

<u>Phthalic anhydride</u> (1) estimated total cost at Ashuganj, including transport from plant at Karachi, using data from Volume III (11) C & F price of invorted material.

| | | | Volume II |
|---|--------------------------------|--|---------------------|
| • | HUMPHREYS & GLASGOV | Y LTD. | 4-34 |
| | UNIDO Vienna for Pakistan & | Pre-Investment Studies for Pertiliser Petrochemical Industries - Final Report | C.1669 July 1970 |

4.2.6 Poreign Exchange Analysis

4.2.6.1 <u>General</u>

The incremental production beyond that of Phase I of the E.P.I.D.C. scheme is shown in the Block Diagram FIG. II 4.1. This is summarised below :-

TABLE II 4.18

INCREMENTAL PRODUCTION FOR PHASE II

| Acetylene | 19,150 |
|-------------------|--------|
| Diectyl Phthalate | 10,000 |
| Acetic Acid | 1,650 |
| Vinyl Acetate | 8,420 |

The basis for the Foreign Exchange Analysis is the comparison of the foreign exchange saving and the local currency cost (excluding duty) for this incremental production, and hence a statement of the cost in local currency of saving foreign exchange. It is applied to the increment in production at the full (design) rate.

4.2.6.2 Variable Costs of Production

From the operating data given in Section 4.2.2 and the unit operating costs given in Section 4.2.5, the variable costs of production for acetylene, acetaldehyde, 2-ethyl hexanol, dioctyl phthalate, acetic acid and vinyl acetate may be calculated.

Phthalic anhydride from indigenous manufacture is assumed. These variable costs are summarised below --

TABLE II 4.19

VARIABLE PRODUCTION COSTS (Basis 1 MT of Product)

| | <u>Poreian Exchanae</u> | Local Cost Excluding Duty |
|-------------------|-------------------------|------------------------------|
| Acetylene | 22.0 | 229.9 |
| Acetaldehyde | 7.1 | 19.7 |
| 2-Sthyl Hexanol | 21.5 | 71.2 |
| Dioctyl Phthalate | 3.2 | 650.0 |
| Acetic Acid | 4.1 | 67.1 |
| Vinyl Acetate | 3.3 | 74.8 |

| NAMENARYS & GLA | GOW LTD, | | | 4-3 |
|------------------------------|--|--|--|--|
| UNIDO Vienna for Pakistan | Pre-Investment Studies fo & Petrochemical Industries | r Ferti - Final | liser Report | C.1669 July 1970 |
| | | | | |
| 1, | | | | |
| | | | | |
| | Notes on Table II 4.19 | _ | | |
| | packages where necessary | bove ta • | ble include th | e cost of |
| | 2. Summation of the variable variable costs of the co Thus, the variable cost appears in the list is e | e costs mplex e: of any p excluded | listed enable xtension to be precursor whic | s the total calculated. h also |
| 4.2.6 | .3 Fixed Costs of Production | | | |
| | Maintenance | | | |
| | For the purposes of this call on the foreign exchange invo for the Fauji development (following basis was used - | loulatio Stment See Sect | on, maintenano in a similar tion 3.2.6.3). | e was based way to that The |
| | Foreign Exchange Element of Maintenance costs | 2 .9% | of foreign e. capital cost | xoh ange |
| | Local Cost Element of Maintenance costs | 2.5% | of foreign en capital cost | xchange |
| | For the total foreign exchanges (see Section 4.2.4), the main | nge inve Intenand | e charge per | 47 x 10 ⁶ \$, year is:- |
| | Foreign Exchange Local Cost (excluding duty) | \$ Ra | 420 ,000 1 ,726,000 | |
| | Labour and Supervision | | | |
| | The direct labour requirement expansion and the other plan ohemicals lists in Table II 107 men (see Section 4.2.2), 9,000 Rs per man per year, w the costs are :- | ats of t ats corr 4.17, s At s high al | the acetylene presponding to the estimated the stimated the stimated the stimated the stimated the stimate stimates are stimated to the stimates and stimates are stimates and stimates are stimates and stimates are | plant the petro- to be t of rvision, |
| | Local Cost | Rs | 963,000 | |
| | Overheads | | | |
| | These are estimated at 3% of This is calculated using the thus :- | the to data f | tal sales reve rom Table II 4 | nue. 16, |
| | | | | |
| | | | | |

jian-

-

· • • • • ·

|) | | BOOW LTD. | Volume II 4-36 |
|---|--------------|---|-------------------|
| | UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| | for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

| | | No x 10 ⁶ |
|--------------|---|----------------------|
| <u>Selee</u> | 16,000 MTPA Dioctyl Phthelate 1,650 MTPA Acetic Acid | 44.8 |
| | 8,420 MTPA Vinyl Acetate | 15.0 |
| | | 61.0 |

The overheads are thus Rs 1,856,000 (local cost only).

Interest on Working Capital

Working capital was assumed to be equivalent to the value of 2 months of finished product. As output rises during the initial years, so also does working capital. The interest on this working capital is equivalent to 1.5% of the years sales revenue. It is thus Rs 928,000 for full production capacity.

The fixed costs of production are summarised below:-

TABLE II 4.20 FIXED PRODUCTION COSTS

| | Fore Lan Stohange | Local Cost Excluding |
|--------------------------------|----------------------------|------------------------------------|
| | (\$ x 10 ³ /yr) | $\frac{Ducy}{(Rs \times 10^3/yr)}$ |
| Maintenance | 420 | 1.726 |
| Labour and Supervision | N11 | 963 |
| Overheads and Selling Expenses | N11 | 1,856 |
| Interest on Working Capital | N11 | 928 |
| TOTAL : | 420 | 5,473 |
| | | |

4.2.6.4 Foreign Exchange Saving and Rupes Cost Comparison

The following table compares the foreign exchange saving and the Rupse costs of production, and is compiled using the data given in the previous sections.

The Rupse element of the investment excludes duty on equipment, but includes start-up expenses. The use of the average interest rate ensures that the early years are not penalised nor the results for the later years over-optimistic.

| | | | ••• • |
|-------------|---|----------------------------------|---------------------------|
| | SOW LTD. | | Volume II 4-37 |
| NIDO Vienna | Pre-Investment Studies for Fe | rtiliser | C.1 66 9 |
| or Pakistan | & Petrochemical Industries - Fi | nal Report | July 1970 |
| • • | | | |
| 4.2. | .0.4 Foreign Exchange Saving and Rup | ee Cost Compar | ison (cont'd.) |
| | | <u></u> | |
| | TYNELON SAURANOE | MALISIS | |
| | | Foreign Exchange | Local Cost Excl. Duty |
| | | \$ x 10 ³ /yr. | Rs x 10 ³ /yr. |
| | Variable Production Costs | | |
| | Acetylene, 19,150 MTPA | 421 | 4,400 |
| | Acetaldehyde, 25,900 MTPA | 184 | 510 |
| | Dioctyl Phthalate. 16.000 MTPA | 805 51 | 10 400 |
| | Acetic Acid, 1,650 MTPA) | مر ط | 10,700 |
| | 6,530 MTPA) Vinvi Acetate 8 400 MTPA | 27 | 5 4 9 |
| | Pierd Destination Contra | | 630 |
| | CANTE FEOGLICTION COSTS | 420 | 5,473 |
| | TOTAL PRODUCTION COSTS | 1.393 | 22,805 |
| | Foreign Exchange Value of Produc | te | |
| | Dioctyl Phthalate, | | |
| | 16,000 MTPA @ \$384 Acetic Acid | 6,140 | |
| | 1,050 MTPA 0 \$161 Vinvi Acetate. | 266 | |
| | 8,420 MTPA • \$222 | 1,870 | |
| | TOTAL F.E. VALUE OF PRODUCTS | 8,276 | |
| | Average Interest over Rivet | | |
| | 10 years 0 8% (= 4% on total F.E. Loan of \$17.72 x 106) | 709 | |
| | Annual Repayments | 1,772 | |
| | TOTAL INTEREST + REPAYMENTS | 2,481 | |
| | Foreign Exchange Savings | 4,402 | |
| | Cost of P.E. saving Ra/S excludi depreciation on Aupee element | 5.2 | |
| | Depreciation @ 10% on Rupee elem | ent | |

١.

Rupee cost, including depreciation cost of P.E. saving Rs/\$ including depreciation

5.8

| -] | And a second | | | |
|-----|--|--|--|--|
| | | L TD , | | Volume II |
| | UNIDO Vienna for Pakistan | Pre-Investment Studies : & Petrochemical Industries | for Fertiliser 9 - Final Repo | C.1669 rt July 1970 |
| | 4.2.6.5 | Foreign Exchange Analysis f | or Each Produ | • * |
| | | The variable and fi intermediate and product hs same basis as in the preced (both operating and financi between the acetylene expan before, the duty and bonus the costs of plants, materi costs include depreciation | exed production we been calculating sections. al) have been sion and the p elements have als and utility on the local of | n costs of each lated, on exactly the The fixed costs suitably distributed process units. As been omitted from ty services. Financial element of fixed capital. |
| | | The production cost fibures so calculated are:- | s and foreign | exchange savings |
| | | Fore | \$ isn Exchange | Rs Local currency |
| | • | Acetylene, per MT | | |
| | | Variable Fixed | 22.0 38.9 | 229.9 111.3 |
| | | Production Cost | 60.9 | 341.2 |
| | 1 | Acetaldehyde, per MT | | |
| | | Acetylene, 0.63 MT Other variable Fixed | 38.4 7.1 29.5 | 215.0 19.7 84.7 |
| | J | Production Cost | 75.0 | 319.4 |
| | 1 | 2-Ethyl hexanol, per MT | | |
| | | Acetaldehyde 1.66 M Other variable Fixed | I 124.5 21.5 37.6 | 530.2 71.2 107.9 |
| | | Production Cost | 183.6 | 709.3 |
| | J | Dioctyl phthelate, per MT | | |
| | 1 | 2-ethyl hexanol, 0.73 MT Other variable Fixed | 134.0 3.2 21.4 | 517.8 650.0 61.5 |
| |] | Production Cost c/f | 158.6 | 1229.3 |
| | I and the second se | | | |

I

i.

| | N LTD. | | Volume II 4 - 39 |
|-------------|--|--------------|---------------------|
| NIDO Vienna | Pre-Investment Studies fo | r Pertil | iser C. 1669 |
| or Pakistan | & Petrochemical Industries | - Final | Report July 1970 |
| | | | |
| | Production cost b/f Cost of alternative | 158.6 | 1229.3 |
| | import (C & F) Foreign exchange | 384 | |
| | saving | 225.4 | |
| | Ratio Rs 1229.3/\$225 | .4 = | <u>Ra 5,4/8</u> |
| | Acetic acid, per MT | | |
| | Acetaldehyde, 0.77 MT | 57.8 | 245.9 |
| | Other variable Fixed | 4.1 34.5 | 67.1 |
| | | | |
| | Cost of alternative | 90.4 | ●▲↓ • (|
| | Foreign exchange | 101 | |
| | saving | 64.6 | |
| | Ratio Rs 411.7/\$64.6 | • | <u>Re 6.4/8</u> |
| | Vinvl ecetate, per MT | | |
| | Acetylene, 0.32 MT | 19. 5 | 109.2 |
| | Other variable | 3.3 | 290.4 71.8 |
| | Fixed | . 38.0 | 109.0 |
| | Production cost | 130.2 | 580.4 |
| | Cost of alternative | | 2 031 + |
| | Import (C & F) Povelen exchange | 222 | |
| | saving | 91.8 | |
| | Ratio Ra 589.4/201.8 | - | Ra 6.4/8 |

١.

| And shading all shade | | MA A B B B B B | |
|-----------------------|---|-----------------------|------|
| MUTURNETO | • | | LVD. |

Volume TT 4-40

| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
|--------------|---|-----------|
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

4.2.7 Cash Flow and D.C.F. Rate of Return

4.2.7.1 <u>General</u>

The sales revenue is based on the pattern of market buildup stated in the Interim Report. Production by the individual plants is only limited to 30% and 90% in the first and second years of operation in view of the fact that existing plants and utilities will be well established. The lower figure of the two (market requirement or production capacity), establishes the output, and thus the sales rate, for a particular year. After 1980, the full production rate can be maintained and sold.

In this analysis, it was decided to use the average and not the marginal cost of acetylene production in establishing the production costs of the products (see Section 4.2.7.2 below). Total costs of the manufacturer are used throughout this analysis so that the D.C.F. rate of return can be determined from the viewpoint of the Pakistan investor.

4.2.7.2 Average Cost of Acetylene

Phase I of the E.P.I.D.C. project at Ashuganj is planned to produce 36,000 MTPA acetylene. Our recommended development would require this capacity to be increased to 55,150 MTPA. For the purposes of analysing Phase II of the project, acetylene is taken as a feed to the new plants proposed, and the average cost of acetylene production is based on the capacity of 55,150 MTPA.

The estimated total cost of acetylene as a feed to the Phase II units is Rs 1,037/MT.

4.2.7.3 Variable Cost of Dioctyl Phthalate, Acetic Acid and Vinyl Acetate Production

Using the average cost of acetylene production from the previous section, the operating data in Section 4.2.2 and the total unit costs of raw materials and utilities from Section 4.2.5, the variable costs of the three sales products were calculated. The variable costs include the cost of all raw materials used, and also the cost of packages. They are summarised below :-

| Dioctyl Phthalate | Re | 1712.4/0 |
|-------------------|----|----------|
| Acetic Acid | Rs | 696.0/M |
| Vinyl Acetate | Ra | 915.6/MI |

| l | Volume II 4.41 | | |
|---|------------------------------|--|---------------------|
| | UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 July 1970 |
| | | | |
| | 4.2.7 | 7.4 Fixed Costs of Production | |

The fixed costs are estimated for the extension project as a whole and are not subdivided between end-products. In the cash flow table and D.C F. calculation, the presence of the full capacity acetylene unit on the site is assumed.

Maintenance

The total annual maintenance charge for the extension project is estimated as a proportion of the fixed capital for the whole extension project. The basis adopted (consistent with Section 4.2.6.3) is 4% of total fixed capital (duties excluded). On this basis, the figure of 4.0% includes allowance for the duty element of materials used in maintenance.

Labour and Supervision

The direct labour requirements for all five plants is estimated to be 102 men, charged at 9,000 Rs per man per year.

Overheads and Selling Expenses

These are estimated to be 3% of the total sales revenue, as in the Foreign Exchange Analysis. They are based on full production rates.

Foreign Exchange Repayments

The repayments are made in equal sums over the first ten years of operation. The total foreign exchange loan for the plants producing the intermediates and sales products (i.e., excluding the Acetylene Plant Extension) is :-

Rs 63.98 x 10⁶

It includes interest on the loan during the construction period and the foreign exchange capital covering start-up expenses (see Section 4.2.4).

4.2.7.5 Other Fixed Costs

Interest on Foreign Exchange Loan

This is taken as 8% of the outstanding loan.



84.10.15 AD.86.07 ILL5.5+10







MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS STANDARD REFERENCE MATERIAL 1010a (ANSI and ISO TEST CHART No. 2) 24 ×

| | 4.42 | |
|--------------|---|-----------|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 |

Interest on Working Capital

As in the Foreign Exchange Analysis, this is based on 1.5% of the total sales revenue. The total sales revenue applicable to the particular year is used.

4.2.7.6 Cash Flows and D.C.F. Rate of Return

The cash inflows, cash outflows and the net cash flows are set out in Table II 4.21. The net cash flow is before tax and depreciation.

A total span of 15 years is covered; year 0 is the reference year in which it is assumed that all capital expenditure is incurred. (The allowance mentioned above for the interest during construction permits this simplification.)

The costs in this table are based on 1970's costs of equipment materials, and labour. (see 3.2.7.6 for detailed comment). The proposed selling prices of the products are on the same basis and were arrived at by making a preliminary calculation (not shown) of the price which would give a reasonable return on the total capital employed, suitably allocated between the products. They are well below the current import prices on a cash-cum-bonus; the C & F prices which would be competitive with the proposed selling prices are shown in Table II 4.17.

The D.C.F. rate of return based on the Rupse investment in the fixed assets is calculated at 16.4%. This D.C.F. level indicates the viability of starting-up the extension in 1976 at the capacities proposed.

For explanation of the capital costs used, see Section 4.2.4.7.

Volume II

4.43

C.1669

HUNNINEYS & GLASGOW LTD.

UNIDO Vienna

for Pakistan

ł

& Petrochemical Industries - Final Report July 1970 Ē 물록집 'B 21.72 B Ĩ 동물장 В SC.72 ľ C В Ĩ 2 35=6 Ĩ 2 Ż В ĩ B 333 5 g 3 £ 3 Ş Ţ ą JIS 3 đ 29 **3** • 1 Ň

Pre-Investment Studies for Fertiliser

| HUMPHREYS & GLASGO | W LTD. | Volume II 4,44 | |
|--------------------|---|-------------------|--|
| UNIDO Vienna | Pre-Investment Studies for Fertiliser | C.1669 | |
| for Pakistan | & Petrochemical Industries - Final Report | July 1970 | |

4.3 Conclusions

1. Consideration of the Phase I plans for production at Ashuganj and our recommendation for Phase II in West Pakistan, indicates that the new products for Phase II at Ashuganj should be:

> 16,000 MTPA Dioctyl Phthalate 8,420 MTPA Vinyl Acetate

1,650 MTPA Acetio Acid

Each of the above products shows a saving on foreign exchange. A scheme including these products, and requiring an expansion of acetylene capacity from 36,000 MTPA to 55,000 MTPA, has been developed and a start-up date of 1976 is proposed.

 \sim . The Rupse cost of saving foreign exchange for the scheme as a whole is estimated at Rs. 4.78/U.S. \$, and is very favourable.

3. Considering acetylene from the expanded acetylene plant as a feedstock, the D.C.F. rate of return on new Rupee capital invested in the plants for the above products has been calculated at 16.4% (before tax).

It may well be thought adequate for a publicly-owned development corporation such as EPIDC and will allow EPIDC to pay the interest rate on the ivestment which will be required by the central authorities.

4. If the D.C.F. rate of return is considered too low by EPIDC the proposed start-up date for the scheme could be delayed. This would increase the rate of return as the utilisation of plant capacity in the first 4 years would be improved. (Under the scheme proposed at present, start-up is in 1976 but capacity is designed to meet the estimated consumption for the products in 1980). Any delay, however, postpones the date when the foreign exchange starts to accrue.

5. If the Phase II scheme proposed here is accepted by EPIDC, then the acetylene plant and associated facilities should be designed at the outset for the full required capacity of 55,000 MTPA acetylene.

6. An export survey should be undertaken of the products proposed for Phase I and Phase II of the EPIDC Petrochemicals Project.

| [| HUMPHREYS & GLA | BOW LTD. | Appendix I Page 1 | |
|---|------------------------------|--|----------------------|--|
| | UNIDO Vienna for Pakistan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 July 1970 | |

. .

APPENDIX I

.

BIBLIOCRAPHY

Reference

| 1. | C. Jenic: UNDP Project Manager General Survey of the Existing Situation in the Petrochemical a Fertilizer Industries, and | nd |
|----|--|----------------|
| | A Review of the Raw Materials Availabilities and their Prices i Pakistan | n May, 1967 |
| 2. | T. Janakievski: UNDP Project Chemical Engineer General Survey of Existing Situation in regard to the Manufactu of man-made fibres in Pakistan. | rer |
| | | March, 1968 |
| 3. | C. Jenic: UNDP Project Manager A Preliminary Study of Complex Fertilisers with Particular refe to Pakistan. | rence |
| | | March, 1968 |
| 4. | T. Janakievski: UNDP Project Chemical Engineer A Review of the Industrial Chemical Complex Study for East Paki and a Proposed Pattern for a Petrochemical Complex. | eten |
| | | ' April, 1968 |
| 5. | V. Tsvetkov: UNTA Expert General Survey of Existing Situation and Proposal for the Plast Production and Processing Industries in Pakistan to 1985. | ice |
| 6. | T. Janakievski: UNDP Project Chemical Engineer A Study of Feedstocks and Processes in the Petrechemical Indust: with a Review of the Position in Pakistan. | ry . |
| | | January, 1969 |
| 7. | L. Kochetkov: UNDP Expert Nethodology and General Survey of the long-term forecast for the Production of Plastics and Synthetic Fibres and background of the Chemical Industry in Pakistan Economy. | e he |
| A | | bruary, 1969 |
| •• | Fertilizer Marketing in Pakistan. | |
| | | October, 1968 |
| 9. | A. Anania: FAO Agricultural Credit Specialist Pointers for Discussion of Credit Problems connected with Perti- Promotion in Pakistan 1968. | lizers |
| | (Superseded by Ref.13) | |

| | | OW LTD. | Appendix I Page 2 |
|---------------------------------|---|---|---------------------------|
| UNIDO V for Pak | 'ienna Listan | Pre-Investment Studies for Fertiliser & Petrochemical Industries - Final Report | C.1669 July 1970 |
| | | | |
| 10. J. | G. Vermaat; | FAO Team Leader | |
| (a) | Prelimina: Compound (| ry Bvaluation of the Prospects of Complex and Complex and Compound Fertilisers in Pakistan. | |
| (b) | Projection for the Pe | n of Fertiliser Nutrients Requirements of West | January, 1968 Pakistan |
| (a) | Tertilizer | | Pobruary, 1968 |
| (0) | F#4 VI112 41 | ose rotential in Mat Pakistan. | March, 1968 |
| 11. Bate Mari | elle Institu Cet Survey d | ite, Frankfurt/Main: Report for UNIDO of Petrochemical Products in Pakistan. | |
| | | | January, 1969 |
| 12. Inve Surv | estment Advi Yey of Plast | sory Centre of Pakistan: Government of Pakis ics Processing Capacity in Pakistan. | tan |
| 13. A. A Cred | nania: FAO lit and Fert | Agricultural Credit Specialist ilizer Promotion in Pakistan. | |
| | | | May , 1969 |
| 14. T. J A St in P | anakievski; udy of Arom akistan. | UNDP Project Chemical Engineer atics and Derivatives with a review of the pos | sition |
| 15. L. K Noon Indu | ochetkov: omical Surv stry and And | UNDP Expert ey of Existing Situation in the World's Petroc alysis of West Pakistan Petrochemical Complex. | hemical |
| 16. Repo in t | rt of the A. he BCAFE Rea | JDC Fact-finding Mission on Petrochemical Indu gion. Report to Asian Industrial Development | stries Council. |
| 17. Dr. Metho of We | Andrew S. de ods of Soil est and Tast | • Endredy: FAO Consultant Analysis (Soil Fertility and Soil Testing Ins t Pakistan). | titute |
| _ | | | June, 1969 |
| 8. C. J Gene: and 1 | enic: UNDP rel Survey o Fertilizers | Project Manager of World Production, Consumption, Internationa Requirements. | l Trade |
| | | | September, 1966 |
| 9. Zinde Pakis | er Internati stan. A Fr | onal Ltd./Technocrats Ltd: Report to Governa eliminary Fuel Study for Pakistan. | ent of |
| | | | 1968 |

| NUMPHREVS | S GLASCOW LTD. | Appendix 1 Page 3 |
|------------------------------|--|--------------------------------------|
| UNIDO Vien for Pakist | na Pre-Investment Studies for Fertiliser an & Petrochemical Industries - Final Report | C.166 9 July 19 7 0 |
| | | |
| 20. Locati Pakist | on and Investment in Nitrogenous Fertilizer Production an (Rotner Report). E.R. Rotner Harvard University | n in West Advisory Service. |
| | | December, 1967 |
| 21. Fertil Direct | izer Policy in Th st Pakistan. Minutes of the Meetin or of Agriculture Th st Pakistan Government. | g. |
| 22. Dr. J. Techni | G. Vermaat: FAO Team Leader cal Reports on Fertilizer Trials (That Pakistan) - Ver | March, 1969 |
| 23. Two Pe | | Louis Reports. |
| (a) C | FOR Bestonses to Vericus Destatis Partilizer Co. Ltd. | |
| | top Responses to Various Phosphatic Fertilizer, West] | Pakistan. |
| (b) ¥ | et Pakistan Fertilizer Compution Pressed | November, 1966 |
| (-) " | Constant for extriber constant for Forecast. | Western 10// |
| 24. Soil F Dr. J.(| ertility Investigations in Pakistan. 3. Vermaats FAO Team Leader | November, 1966 |
| | | F.A.0. 1964 |
| 25. Soil Fe Von Pet | ertility Investigations in the Peshawar Region. For FAO Consultant | |
| | | F.A.0. 1964 |
| 26. The Fer Plannin | tilizer Programme of the Government of Inst Pakistan. ag & Development Dept. That Pakistan Government. | |
| 27. Fertili Dr. J.(| zer Requirements for East Pakistan (1970-1980). 1. Vermaat: FAO Team Leader | |
| 28. West Pa U.S. Ag | kistan Chemical Fertilizer Handling and Marketing Sur ency for International Development - Pakistan. | vey. |
| 29. The Dem West Pa | and for Fertilizer in West Pakistan, <u>WPADC</u> . kistan Agricultural Development Corporation. | |
| | | December, 1968 |
| 30. The Man | ufacture of Fertilizers in Tast Pakistan (3 volumes) | |
| Vol. I Vol. II Vol. II | Survey of Demand Manufacture of Nitrogenous Fertilizer. See Ref. I Manufacture of Phosphatic Fertilizer. | 31. |
| Chemoon | Report to Industrial Managements Ltd. | |
| | | August, 1967 |

| n | | Annendi - Y |
|---|---|------------------------------|
| U | HUMPHNEYS & GLASGOW LTD. | Appendix 1 Page 4 |
| I | UNIDO Vienna Pre-Investment Studies for Pertiliser for Pakistan & Petrochemical Industries - Final Report | C. 1669 Ju ly 1970 |
| | | |
| | 31. New Incue of (30) Vol. I and II only. | |
| l | 32. Bonomic and Technical Braluation of Overseas Shipmont and Util of Elemental Phosphorous for Pertiliser Production. Tennessee Authority. Published Report. | isation Valley |
| | | January, 1968 |
| ſ | 33. The Bole of Extension in the Distribution of Pertilisers. Dr. J.G. Vermaat. FAO Team Leader | |
| - | 34. Seminar on Petroleum. Petroleum Institute of Pakistan. Se | ptember , 1968 |
| [| 35. Sthylene and Polyethylene Plant for Inst Pakisten. Mitsubishi. Report to Inst Pakistan Industrial Development Corporation. D | ecember, 1966 |
| l | 36. Market Survey and Feasibility Study for Karachi Petrochemical C (FAUJI). BEICUP & Chemical Consultants (Pakistan) Ltd. Report Foundation. | omplex to Pauji |
| I | 37. Petrochemical Investment in S.E. Asia - French Institute of Pet V.E. Henny F. Vaches Institut Francaie du Petrole. | rol. |
| I | 38. Development of Petrochemical Industry in Pakistan. (2 volumes) Library. Nessrs. Imhausen. Report to Pakistan Qove | rnment |
| I | 39. Development of Fertilizer Industry in Pakistan. | 1964 |
| | Acaers. Indeusen. Report to Pakistan Upvernment. | |
| | eve reu a rea fibre - mribc/sumitomo. Report to S. Pakistan Indj. | Devt.Corp. |
| ľ | 41. PACN R.C.D. Report-Manufacture of Acrylic Fibre Iran - Pakietan Chemical Consultants (Pak) Ltd. Report to W. Pakistan Indl. Dev | t. Corp. 1968 |
| | 42. Pertiliser Trials in Farmers' Fields by A. Wahnab T.I. Department | nt of |
| T | Agriculture west rakistan (byernment. No | vember, 1965 |
| - | 43. Ihst Pakistan Market Survey - INSO Pakistan Pertiliser Company : | Ltd. |
| 1 | | August, 1968 |

| UNID for | 0 Vienna Pro-Investment Studies for Fertiliser Pakistan & Petrochemical Industries - Final Report | C.1669 July 1970 |
|--------------|--|----------------------|
| 4. | Proposal of Peasibility Pattern of Petrochemical Complex Ind Pakistan. Vol III. T. Janakievski. UNIDO Project Chemical | ustry in Ingineer |
| | | June, 1969 |
| 451. | Investment Study. That Pakistan Petrochemical Complex based Chas. Austrian Petrochemical Consultants. Report to That Paki Ind. Devt. Corp. | on Natural stan |
| | | September, 1969 |
| 15 B. | Synopsis of Document 45. | |
| 51. | West Pakistan. Comments on the fertiliser programme during to five year plan. Dr. N. Neumann UNIDO Project Manager | he fourth |
| | | August, 1969 |
| 52. | The Consumption and production of fertilisers in That Pakist during the fourth five year plan. Dr. J.G. Vermaat PAO Teau | an n Leader |
| | 8 | eptember, 1969 |
| 54. | Paper presented in Teheran on the fartiliser industry in Pake (including posticides) Expert from Pakistan Covernement. | isten |
| | | October, 1969 |

•
SOME FIGURES OF THIS DOCUMENT ARE TOO LARGE FOR MICROFICHING AND WILL NOT BE PHOTOGRAPHED.

8-535

84.10.15 AD.86.07 ILL5.5+10